

# GOPEN ACCESS

**Citation:** Keo B, Li B, Younis W (2025) Measuring trade costs and analyzing the determinants of trade growth between Cambodia and major trading partners: 1993–2019. PLoS ONE 20(1): e0311754. https://doi.org/10.1371/journal.pone.0311754

Editor: Imran Ur Rahman, Leshan Normal University, CHINA

Received: November 2, 2023

Accepted: September 24, 2024

Published: January 24, 2025

**Copyright:** © 2025 Keo et al. This is an open access article distributed under the terms of the <u>Creative</u> <u>Commons Attribution License</u>, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data supporting the findings of this study are obtained from various public sources and can be openly accessed through the respective URLs provided for each variable as follows. The primary source of the bilateral exports comes from the Direction of Trade Statistics (DOTS) database of the International Monetary Fund (IMF) [https://data.imf.org/?sk= 9D6028D4-F14A-464C-A2F2-59B2CD424B85]. Supplementary sources for the bilateral exports are also derived from the Organisation for Economic Co-operation and Development (OECD)'s Structural Analysis (STAN) Databases [https:// RESEARCH ARTICLE

Measuring trade costs and analyzing the determinants of trade growth between Cambodia and major trading partners: 1993–2019

# Borin Keo [] \*, Bin Li\*, Waqas Younis

School of Economics and Trade, Hunan University, Changsha, China

\* keo.borin@yahoo.com (BK); libin43@sohu.com (BL)

# Abstract

High trade costs pose substantial barriers to the process of trade liberalization. This study aims to measure trade costs and explore the driving forces behind the growth of bilateral trade between Cambodia and its top 30 trading partners from 1993 to 2019. Using a microfounded measure of trade costs derived from the gravity model, we find that Cambodia's average trade costs decreased by 35.43 percent between 1993 and 2019. Fluctuations in average trade costs persisted until 2014, despite Cambodia's accession to the World Trade Organization (WTO) in 2004. Since then, these costs have declined more rapidly. Cambodia's bilateral trade costs are lower with its major trading partners in Southeast Asia and East Asia than with those in South Asia, Oceania, Europe, and North America. Cambodia's average trade costs with developing and emerging economies are lower than those with developed economies. Between 2014 and 2019, Cambodia experienced a notable decline in average trade costs with trading partners along the Belt and Road Initiative (BRI) corridors by 34.78 percent, twice as fast as with non-BRI trading partners. Regarding the decomposition of trade growth, we find that the expansion of Cambodian trade over the period from 1993 to 2019 was driven by three factors: the rise in income (59.65 percent), the decline in trade costs (56.69 percent), and the decline in multilateral resistance (-16.34 percent). The findings of this study have significant implications for a better understanding of Cambodia's development process toward global trade integration over the past two decades. Our results suggest that Cambodia can optimize its trade expansion potential by focusing on its relations with trading partners exhibiting high economic growth potential and those showing substantial reductions in trade costs.

# 1. Introduction

Trade costs have long been of great interest to trade economists owing to their critical roles in economic growth [1, 2], trade expansion [3, 4], trade networks [5], foreign direct investment (FDI) [6, 7], national welfare [8, 9], and economic development [9, 10]. In a broad sense, trade

stats.oecd.org/Index.aspx?DataSetCode=BTDIXE\_ I4#] and the United Nations Conference on Trade and Development (UNCTAD) database [https:// unctadstat.unctad.org/datacentre/dataviewer/US. TradeMatrix]. All domestic trade data are available from the International Trade and Production Database for Estimation Release 2 (ITPD-E-R02) [https://www.usitc.gov/data/gravity/itpde.htm]. All Gross Domestic Product (GDP) data are accessible via the IMF's World Economic Outlook (WEO) database [https://www.imf.org/en/Publications/ SPROLLs/world-economic-outlook-databases].

**Funding:** The author(s) received no specific funding for this work.

**Competing interests:** The authors have declared that no competing interests exist.

costs refer to all costs paid by consumers to obtain goods in addition to production costs [10]. According to the seminal Anderson and van Wincoop study [10], the tariff-equivalent trade costs of industrialized countries might be as high as 170 percent. This number can be further disaggregated into 55 percent for local distribution costs and 74 percent for international trade costs. Hummels et al. [11] argue that reducing a country's trade costs can increase vertical specialization and trade volumes, ultimately improving a country's position in global value chains (GVCs). Jacks et al. [4] found that a 16 percent decline in trade costs between 1950 and 2000 contributed almost 31 percent to the growth of global trade. These findings imply that despite increasing economic integration, trade costs continue to exert a profound impact on trade. In recent years, international organizations such as the World Trade Organization (WTO) and the Organization for Economic Co-operation and Development (OECD) have placed trade costs on their agendas. This inclusion is due to the significant global policy relevance of trade liberalization and trade facilitation for developing countries, especially least-developed countries (LDCs) [12].

Like in many other LDCs, high trade costs have long been a major obstacle to Cambodia's trade integration and liberalization. Inadequate infrastructure and inefficient transport and logistics facilities are the fundamental factors contributing to these high trade costs. Recent research indicates that Cambodia has the highest additional trade costs for inputs in specific industries, including textiles, chemicals, and computers, compared to other developing countries [13]. As of 2021, Cambodia's total imports comprised 59.87 percent of intermediate goods and 23.96 percent of consumer goods, compared to 19.87 percent of intermediate goods and 20.41 percent of consumer goods in the East Asia-Pacific region [14]. The fact that Cambodia heavily relies on imported intermediate and final goods implies that its high trade costs are likely to diminish the economic well-being of producers, suppliers, traders, and consumers.

On the policy side, the issue of trade costs has also been acknowledged in Cambodia's national policy agendas. For example, as highlighted by the Cambodia Trade Integration Strategy (CTIS) (2019–2023) in 2019 and the recent Pentagonal Strategy Phase-I (2023–2028) in 2023, high trade costs have been recognized as prominent trade barriers to be addressed in order to enhance Cambodia's international competitiveness and promote the diversity of its exports [15, 16]. The significance of export diversification and international competitiveness to support Cambodia's ambition to become an upper-middle-income country by 2030 and a high-income country by 2050 has been emphasized in many of its national policies and strategies. These include the Industrial Development Policy (IDP) (2015-2025) in 2015, the Rectangular Strategy-Phase IV (2018-2023) in 2018, the Cambodia Trade Integration Strategy (CTIS) (2019–2023) in 2019, the Economic Diplomacy Strategy (2021–2023) in 2021, and the Pentagonal Strategy-Phase I (2023-2028) in 2023, among various others. The literature suggests that high trade costs pose major challenges for a country to leverage local comparative advantages for export competitiveness [17], facilitate export diversification [18, 19], and engage in vertical specialization [11, 20]. For instance, Mora and Olabisi [19] found that high trade costs hinder developing countries' capacity to expand their new product lines or boost the export value of their current trade links. It is clear that reducing trade costs in Cambodia can play an indispensable role in its trade growth, economic integration, vertical specialization, export diversification, and international competitiveness.

Despite the extensive evidence highlighting the profound significance of trade costs in Cambodia from a practical and policy standpoint, little is known about their magnitudes, evolutions, and consequences. So far, the existing research on trade costs has made substantial contributions to deepening our understanding of their scales, determinants, and impacts [4, 10, 21]. However, to our knowledge, there have been hardly any empirical studies that

exclusively examine trade costs in Cambodia. One of the main reasons for this is the difficulty in collecting reliable data on measuring trade costs, either directly or indirectly. While few studies have explored the factors affecting bilateral trade flows between Cambodia and its major trading partners using the traditional gravity model [22, 23], there has been a lack of an in-depth analysis of trade costs. For instance, using the traditional gravity model, Huot and Kakinaka [23] investigated the factors influencing Cambodia's bilateral trade flows with its top 20 trading partners from 2000 to 2004. Their study found that Cambodia's trade volumes decreased between 0.9 and 1.4 percent for every percentage point of increasing distance from its trading partners. The authors employed distance as a proxy for transport costs, which only captured a subset of trade costs. Therefore, the overall trade costs that have impeded Cambodia's trade flows might not be accurately measured. According to Anderson and van Wincoop [10], the overall trade costs cover all costs paid by consumers to obtain goods in addition to production costs. They include various elements, such as tariffs and non-tariff barriers, transport costs, border-related costs, language and communication barriers, and different currency effects, among others. In addition, as emphasized by Novy [24], conventional measures, such as distance and other static proxies for trade costs, cannot reflect the evolutions of trade barriers over time. These limitations in the literature highlight the need for an alternative approach that can determine Cambodia's comprehensive trade costs.

In light of the above discussions, this study aims to measure trade costs in Cambodia and explore the driving factors of its trade growth. To elaborate further, this research seeks to answer the following questions: What has been the trend in trade costs between Cambodia and its major trading partners over the past two decades? Which countries or regions have witnessed the most rapid decrease in their bilateral trade costs with Cambodia? What are the magnitudes of the remaining impediments? Are there any differences between Cambodia's average trade costs with developing countries and those with developed countries? Has the Belt and Road Initiative (BRI) reduced trade costs between Cambodia and its major trading partners? What factors have contributed to the expansion of Cambodia's trade? Is it due to changes in income or trade costs? To answer these questions, we employ a micro-founded method of trade costs introduced by Novy [24] to measure Cambodia's trade costs with its top 30 trading partners from 1993 to 2019. Unlike prior research in Cambodia that has solely relied on geographic distance as a proxy variable for transportation costs, this method allows us to track all costs that hinder the Cambodian trade flows without making assumptions about the individual components of these costs. We also apply the decomposition model of trade growth to examine the driving determinants of the Cambodian trade expansion during the same period.

This study contributes to the existing literature on trade costs in several ways. First, a deeper understanding of trade costs from LDCs, particularly low-income or lower-middle-income countries, has been limited [25–27]. Prior studies in this area have mainly focused on devel-oped countries [24, 28], newly industrialized countries [29, 30], and a group of major devel-oped and emerging economies [31, 32]. As highlighted by Anderson and van Wincoop [10], trade costs are much higher in developing countries and vary across countries. Therefore, this study offers new insights into how a least-developed country can mitigate trade barriers. In addition, earlier research findings on trade costs in developing countries have been inconclusive. For instance, some studies have shown a decline in trade costs over time [30, 33], whereas others have revealed no changes [29, 34, 35]. These conflicting results in the literature underscore the importance of conducting empirical studies to gain a better understanding of trade costs in developing countries that have also embraced trade liberalization. Next, the unique contribution of this research is that it is the first attempt to measure Cambodia's trade costs. Finally, this study presents novel data covering the 27-year period from 1993 to 2019, including Cambodia's top 30 trading partners, which accounted for nearly 94 percent of its overall

trade volumes as of 2019. This extensive dataset allows us to track the dynamic changes in Cambodia's trade costs over time, from the initial stages of its economic reform in the early 1990s until the recent period, reflecting the country's trade openness with the world. By analyzing this data, we can also investigate the driving forces behind the growth of the Cambodian trade over the past two decades. Therefore, the findings will provide policymakers with the necessary information to target their policies to reduce trade costs and promote trade liberalization.

Our study presents multiple pieces of new evidence. First, Cambodia's average trade costs decreased by 35.43 percent between 1993 and 2019, with a remarkable decline of 24.66 percent between 2014 and 2019 alone. There were variations in these costs until 2014, even though the country joined the WTO in 2004. Next, Cambodia has lower bilateral trade costs with its major trading partners in Southeast Asia and East Asia than with those in South Asia, Oceania, Europe, and North America. Cambodia's average trade costs with developing and emerging economies are lower than those with developed economies. In terms of the magnitudes of these changes, Cambodia's average trade costs with developing and emerging economies have declined at twice the rate of those with developed economies since 2014. Notably, between 2014 and 2019, Cambodia's average trade costs with trading partners along the Belt and Road Initiative (BRI) corridors fell by 34.78 percent, double the rate of those with non-BRI trading partners. This finding aligns with the literature indicating that countries or regions along the BRI corridors where transport infrastructure projects have been developed experience a considerable decline in trade costs by up to 10.2 percent [36]. We also find that the expansion of Cambodian trade between 1993 and 2019 was driven by the rise in income (59.65 percent), the decline in trade costs (56.69 percent), and the decline in multilateral resistance (-16.34 percent). This study highlights the significance of reducing trade costs in supporting Cambodia's efforts to diversify its exports and enhance its international competitiveness. This will help Cambodia achieve its ambition to transform from a least-developed country to an upper-middle-income country by 2030 and a high-income country by 2050.

The remainder of this paper is outlined as follows. Section 2 provides an overview of Cambodia's trade development. Section 3 reviews the literature, presenting the theoretical background and empirical studies on trade costs. Section 4 contains the methodology for measuring trade costs and decomposing trade growth factors. Section 5 provides the empirical findings and discussions for Cambodia. Section 6 concludes. Finally, the <u>S1 Appendix</u> describes our data and sources.

# 2. Overview of Cambodia's trade development

Since adopting a market-based economy in 1993, Cambodia has implemented a series of major reform policies and strategies to boost exports, which are essential to economic development, employment, and poverty alleviation. One of the early national government policies was the first five-year Socio-Economic Development Plan (SEDP) of 1996–2000. Its primary goals were to achieve "high growth, employment generation, and poverty reduction", which were accomplished by promoting rural and labor-intensive industries through an export-led growth strategy [37]. Cambodia launched its first Diagnostic Trade Integration Strategy (DTIS) in 2001 after becoming a member of the Association of Southeast Asian Nations (ASEAN) in 1999. The adoption and implementation of these trade strategies made Cambodia the first LDC to join the WTO in 2004 through the complete accession procedure.

Consequently, Cambodia has witnessed remarkable growth in its trade. For example, Cambodia's exports and imports of merchandise increased from US\$ 284 million and US\$ 471 million in 1993 to US\$ 14.82 billion and US\$ 20.28 billion in 2019, respectively (see Fig 1). From



Fig 1. Cambodia's merchandise trade volumes from 1993 to 2019 (in million USD). Source: Authors' calculation using the merchandise export and import volumes, which are derived from the World Trade Organization (WTO) Stats data portal (https://stats.wto.org/).

https://doi.org/10.1371/journal.pone.0311754.g001

1993 to 2019, Cambodia recorded a 46.5-fold increase in its total merchandise trade volumes, rising from US\$ 755 million to US\$ 35.1 billion, equivalent to a growth rate of 15.9 percent per year. In comparison, the annual average growth rate of trade in goods in the ASEAN and the world during the same period stood at 7.3 percent and 6.3 percent, respectively. The average annual growth rate was computed by applying the compound annual growth rate from 1993 to 2019 for the total merchandise trade (export plus import) volumes in nominal terms, using the data from the WTO database [38]. It is evident from the data that Cambodia's trade witnessed an impressive growth rate over the period from 1993 to 2019, outperforming both the ASEAN and the global trade by more than double. Simultaneously, Cambodia's merchandise trade as a share of its gross domestic product (GDP) increased from 29.8 percent in 1993 to 129.6 percent in 2019 [39]. In 2016, Cambodia became the second-largest exporter of manufactured goods among the LDCs after Bangladesh, with total exports reaching more than US\$ 10 billion [40]. This rapid growth in Cambodian trade over the past two decades can be attributed to various factors, including a decrease in both tariff and non-tariff barriers, the signing of more regional trade agreements (RTAs), improvements in information and transport technology, better infrastructure development, a growth in foreign direct investment (FDI), and an increase in income. Previous studies have verified that reducing trade costs is necessary for expanding international trade [3, 4, 24]. For instance, Jacks et al. [4] found that the growth of global trade between 1950 and 2000 was contributed by nearly 31 percent due to a decline in trade costs.

Cambodia had been recognized as one of the most rapidly growing economies in the world until the COVID-19 pandemic hit in 2020. The Cambodian economy grew by 7.7 percent annually on average between 1995 and 2019, while the per capita income increased from US\$

323 in 1995 to US\$ 1,621 in 2019, allowing Cambodia to attain its lower-middle-income status in 2015 [41]. This remarkable economic development in Cambodia over the past 25 years would not have been possible without the rapid growth of its trade. Despite all these achievements, Cambodia's exports have been concentrated largely on low-value-added products, including textiles, clothing, footwear, travel accessories, and rice. Over the past decade, more than half of Cambodia's exports in goods have gone to the United States and the European Union (EU) markets [42]. This heavy dependence on a few developed countries makes Cambodia's exports vulnerable to changes in their market demand. As an illustration, Cambodia recorded a decline in exports in 2009 because of the 2008–2009 Global Financial Crisis and again in 2013 and 2014, partly due to the low import demand in developed countries and the recession in the European Union (see Fig 1) [43]. The other concern that needs to be considered is that if Cambodia graduates from its LDC status, it will no longer be eligible for trade preferences or measures related to international assistance under the Generalized System of Preferences (GSP).

The empirical investigation has demonstrated that Cambodian exports would decrease by up to 10 percent upon transitioning from its LDC status [44]. It is clear that the conventional approach of depending on exports of low-value-added products to boost Cambodia's economic growth is no longer sustainable. Therefore, a new development paradigm is necessary to diversify Cambodia's export markets or upgrade its current low-value-added products to medium- or high-value-added products. In this sense, the Royal Government of Cambodia (RGC) is fully aware of these challenges and has actively integrated the country's trade and economic activities through various bilateral, regional, and global measures, such as trade facilitation initiatives, economic partnership agreements, and regional trade agreements (RTAs). These include the Belt and Road Initiative (BRI) proposed by China in 2013, the Cambodia-Hong Kong (China) Free Trade Agreement in February 2021 under the ASEAN-Hong Kong (China) Free Trade Agreement (AHKFTA), the Regional Comprehensive Economic Partnership (RCEP) in January 2022, the Cambodia-China Free Trade Agreement (CCFTA) in January 2022, the Cambodia-Republic of Korea Free Trade Agreement (CKFTA) in December 2022, and the Cambodia-United Arab Emirates Comprehensive Economic Partnership Agreement (CAM-UAE CEPA) in 2023. It is important to note that the RCEP, officially coming into effect on January 1, 2022, is the world's largest free trade agreement (FTA) in terms of its economic and population size [45]. This trade block comprises 15 countries in the Asia-Pacific regions, including five countries of the ASEAN Plus Six (Australia, China, Japan, New Zealand, and the Republic of Korea) and ten countries of the ASEAN (Brunei, Cambodia, Indonesia, Laos, Malaysia, Myanmar, the Philippines, Singapore, Thailand, and Vietnam). In recent years, Cambodia has actively tried to negotiate and propose several economic partnership agreements and RTAs with Canada, India, Japan, Switzerland, and the Eurasian Economic Union (EAEU). These efforts aim to reduce Cambodia's trade costs with its major trading partners and diversify its export markets.

# 3. Literature review

# 3.1 Theoretical review of trade costs

This section presents an overview of the theoretical background of trade costs and their measurement methods. Since Paul A. Samuelson [46] introduced the concept of modeling transport costs as "iceberg transport costs" in 1954, trade costs have received significant attention from trade economists in determining trade volumes. However, the role of trade costs in shaping trade patterns has not always been a prominent focus in traditional trade theories. As Deardorff [47] emphasized, many of the earlier trade theories assumed the absence of trade costs, which led to an incomplete understanding of the factors influencing the trade volumes of a country. As international trade theories have evolved, trade costs have emerged as crucial components in prominent theories, such as the new trade theory [48, 49], the new economic geography theory [50, 51], and the theory of heterogeneous firms [52, 53]. Trade costs are also critical in resolving major empirical puzzles within international macroeconomics, such as the home bias in trade, the consumption correlations puzzle, and the purchasing power parity (PPP) puzzle [54]. We adopt a comprehensive definition of trade costs proposed by Anderson and van Wincoop [10]: "Trade costs, broadly defined, include all costs incurred in getting a good to a final user other than the marginal cost of producing the good itself: transportation costs (both freight costs and time costs), policy barriers (tariffs and nontariff barriers), information costs, contract enforcement costs, costs associated with the use of different currencies, legal and regulatory costs, and local distribution costs (wholesale and retail)". According to Jacks et al. [3], trade costs are related to transactions and transportation costs that hinder international economic integration in exchanging goods across borders.

There are two approaches to measuring trade costs: direct measurement [55, 56] and indirect measurement [10, 57, 58]. The direct measurement method mainly selects specific indicators to represent trade costs, such as transportation costs, information costs, tariffs, and nontariff barriers (see Chen and Novy [59] for a brief discussion on the measurement method of trade costs). Although direct measures are straightforward, they are constrained by the unavailability of data, especially for developing countries. They can only be used to quantify specific subsets of trade costs instead of the overall trade costs. Measuring the overall trade costs is challenging since "direct measures are remarkably sparse and inaccurate" [10]. Given the limitations of direct approaches in measuring other components of trade costs, scholars have opted for indirect methods through price data and trade volumes (see [10] for a full review of the trade cost measure using price data). The standard gravity model, the workhorse trade model pioneered by Jan Tinbergen in 1962 [60], has been traditionally used to infer trade costs indirectly from trade flows, with particular variables like geographic distance as a proxy for transportation costs. Other commonly used proxy variables to measure different types of trade costs are currency barriers [61], transportation and infrastructure barriers [55, 56, 62], tariff barriers [62], and language and communication barriers [55, 63, 64], among others. Other scholars have used border-related trade barriers to quantify trade costs [57, 58, 64-66]. Overall, there are two issues with all these measuring techniques. First, they might not entirely reflect all barriers that hinder trade because they cannot capture other hidden trade cost variables, such as cultural differences, institutional barriers, and bureaucratic red tape, which can be difficult to measure using proxy variables. Second, researchers typically determine variables related to trade costs beforehand and incorporate them into the gravity model for regression analyses, which can lead to biased results due to omitted variables. To fill these theoretical gaps, Novy [24], based on Head and Ries [66], developed a micro-founded equation of trade costs that can capture all costs associated with exchanges of goods between two countries or regions without making assumptions about the components of these costs.

#### 3.2 Empirical studies on trade costs

In this section, we present a review of empirical studies on trade costs to highlight the importance of our contribution to the existing literature. In terms of contributions to empirical studies on trade costs, Anderson and van Wincoop [10] estimated trade costs in developed countries, and their findings revealed that the tariff equivalent of trade costs amounted to approximately 170 percent. This number can be further disaggregated to 21% transportation costs, 44% border-related trade costs, and 55% sales (retail and wholesale distribution) costs. Novy [24] measured trade costs between the United States and its major trading partners from 1970 to 2000, and the study found that the average tariff equivalent trade costs of the United States declined by 40 percent throughout the sample period. According to the same survey, the average trade costs of the 13 OECD countries decreased by about one-third between 1970 and 2000. Subsequently, the literature has extensively adopted Novy's trade cost equation to estimate comprehensive trade costs between countries or regions. For example, Wen et al. [34] used Novy's trade cost model to measure the bilateral trade costs of China's agricultural goods with its five major trading partners from 1995 to 2007. Their findings revealed that China's trade costs in agricultural goods had not decreased. Similarly, Miroudot et al. [33] employed an identical approach to estimate trade costs in goods and services, and the authors found that trade costs were much higher in services than goods. Their study also indicated a decrease of approximately 15 percent in global aggregate trade costs in goods between 1995 and 2007.

Arvis et al. [27] adopted the same metric to analyze trade costs in manufacturing and agricultural goods in developing countries from 1996 to 2010. The authors concluded that agricultural trade costs were twice as high as those in manufactured goods and were extremely high for low-income countries. According to the same study, only upper-middle-income countries could reduce trade costs faster among developing countries, while trade costs in low-income countries were still very high. Gaurav and Mathur [29] analyzed the bilateral trade costs between India and the European Union trading partners from 1995 to 2010. Their study revealed that average trade costs between India and the European Union steadily decreased until 2001. After that, these trade costs remained stable at around 50 percent for nearly a decade. Turkson [35] also employed Novy's equation to estimate trade costs in 39 sub-Saharan African countries from 1980 to 2003. The author discovered that Sub-Saharan Africa had the highest average trade costs at 271.5 percent, compared with other regions. In addition, Sub-Saharan Africa had not experienced a decline in trade costs during the same period.

In summary, the existing literature on trade costs has the following characteristics. First, previous studies in this area have primarily focused on developed countries, newly industrialized countries, and groups of developed and emerging countries. However, there has been limited research on trade costs in developing countries, particularly the LDCs. Second, early research findings from developing countries have shown significant contradictions. For instance, some studies have discovered a decrease in trade costs over time, whereas others have revealed no changes. Overall, the existing literature has concluded that trade costs tend to decrease among major developed nations. However, this trend does not apply to developing countries, especially the LDCs, where trade costs remain very high or show no signs of decline.

Being the first LDC to join the WTO in 2004, Cambodia's endeavor to reduce its trade barriers has yet to be fully understood, primarily from the perspective of trade costs. Reducing trade costs is even more critical and urgent for LDCs like Cambodia, seeking to gain comparative advantages and increase export competitiveness regionally and globally. Deardorff [17] highlighted that the local comparative advantages determine the competitiveness of export commodities. However, when trade costs increase, the local comparative advantages decrease, thereby restricting the country's ability to enhance its export competitiveness. The importance of trade costs emphasizes the need for an in-depth understanding of their scales, evolutions, and impacts. Unfortunately, there have been hardly any empirical studies that exclusively examine trade costs in Cambodia. Earlier empirical studies on Cambodian trade were conducted by scholars like Kim [22] and Huot and Kakinaka [23], who were among the first to use the traditional gravity model to measure bilateral trade flows between Cambodia and its major trading partners. Despite the valuable contributions of their studies in understanding Cambodia's trade openness before and after its membership in the ASEAN (1999) and the WTO (2004), the investigation of Cambodia's trade costs, including their magnitudes, evolutions, and consequences, has remained largely unexplored. For instance, using the traditional gravity model, Huot and Kakinaka [23] examined the factors influencing Cambodia's bilateral trade flows with its top 20 trading partners from 2000 to 2004. Their study found that Cambodia's trade volumes decreased between 0.9 and 1.4 percent for every percentage point of increasing distance from its trading partners. The authors employed distance as a proxy variable for transportation costs, which might only reflect a portion of the overall trade costs. Therefore, the comprehensive trade costs that hinder Cambodia's trade flows, such as borderrelated costs, logistics costs, and costs associated with trade compliance, can not be fully captured. Moreover, the gravity model can only explain how distance affects the Cambodian trade volumes. However, it does not provide details about the scales of transportation costs or how these costs have evolved. As Novy [24] pointed out, time-invariant proxy variables, such as distance, are not well-suited to reflect the dynamic variations of trade costs over time. Finally, the traditional gravity model does not take into account the impact of multilateral resistance on trade, leading to biased results [58].

Against this background, this study aims to fill these gaps in the existing literature on trade costs in the context of LDCs, particularly in the case of Cambodia. Using Novy's micro-founded equation of trade costs, this study measures Cambodia's trade costs with its top 30 trading partners from 1993 to 2019. We also employ the trade growth decomposition model to investigate the driving forces behind Cambodia's trade growth during the same period. This study provides fresh insights into how Cambodia's trade costs have evolved over the past two decades after the country transformed into a market-based economy in the early 1990s.

# 4. Material and method

# 4.1 Micro-founded measure of trade costs

**4.1.1 Measurement method of bilateral trade costs.** Building upon the famous border trade puzzle between the United States and Canada presented by McCallum [57], Anderson and van Wincoop [58] made a seminal contribution to the gravity model by including a multilateral resistance component and addressing potential bias resulting from omitted variables. The authors propose that the bilateral trade cost equation between countries or regions is a mathematical function of border costs and geographic distance between trading partners. They define the bilateral trade costs  $(t_{ij})$  as a log-linear function  $t_{ij} = b_{ij} d_{ij}^{\rho}$ , where  $b_{ij}$  denotes the border-related variables ( $b_{ii} = 1$  if both regions i and j are in the same country; otherwise,  $b_{ii}$  is the sum of one and the tariff equivalent of the borders separating the regions' respective countries);  $d_{ii}$  represents distance between the two countries or regions, and  $\rho$  is the elasticity of distance. First, since trade barriers vary across countries, Novy [24] raises concerns about the oversimplification of Anderson and van Wincoop's model by assuming that trade costs between countries are symmetric ( $t_{ii} = t_{ii}$ ). Next, the trade cost function might not be accurately defined because it might exclude significant trade cost factors like tariffs. Finally, he emphasizes that static measures, such as geographic distance, cannot capture the variations in trade costs over time. To overcome these shortcomings, Novy fills theoretical and analytical gaps by deriving a micro-founded equation of trade costs based on Anderson and van Wincoop's gravity model. Novy's proposed measure of bilateral trade costs has several major advantages. First, it can calculate all the costs involved in trading goods between two countries or regions without making assumptions about the components of these costs. Second, this method applies not only to cross-sectional data but also to time series and panel data. Third, the micro-founded measure of trade costs exhibits robust theoretical foundations and can be derived from other influential trade models, such as the Ricardian model of comparative advantage [64] and the heterogeneous firms' models [67, 68]. Inspired by these substantial

improvements over the previous models, we follow Novy's equation for measuring the trade costs of Cambodia.

The following gravity equation, derived by Anderson and van Wincoop [58], has the form illustrated in Eq (1):

$$X_{ij} = \left(\frac{Y_i Y_j}{Y^W}\right) \left(\frac{t_{ij}}{\Pi_i P_j}\right)^{1-\sigma}$$
(1)

Where  $X_{ij}$  denotes the nominal export volume from country *i* to country *j*;  $Y_i$  and  $Y_j$  represent the nominal income of countries *i* and *j*;  $Y^W$  is the world's nominal income, defined as  $Y^W \equiv \sum_j Y_j$ ;  $t_{ij} \ge 1$  denotes the bilateral trade costs (an ad valorem term plus one), and  $\sigma > 1$  is the elasticity of substitution between commodities.  $\Pi_i$  and  $P_j$  represent the price indices of country *i* and country *j*, respectively. Anderson and van Wincoop call the two terms "multilateral resistance" to describe these price indices, which can be defined as average trade costs with all trading partners.  $\Pi_i$  is the outward multilateral resistance of country *i*, while  $P_j$  is the inward multilateral resistance of country *j*.

Considering the symmetry of countries *i* and *j* in Eq (1), the gravity equation of the bilateral export from country *j* to country *i*, denoted as  $X_{ji}$ , can be written:

$$X_{ji} = \left(\frac{Y_j Y_i}{Y^W}\right) \left(\frac{t_{ji}}{\Pi_j P_i}\right)^{1-\sigma}$$
(2)

This study refers to the method proposed by Head and Ries [66] and Novy [24] to eliminate multilateral resistance variables.

Following Novy's proposal, we extend Eq (1) to express the domestic trade volumes of country  $i(X_{ii})$ :

$$X_{\rm ii} = \left(\frac{Y_i Y_i}{Y^W}\right) \left(\frac{t_{\rm ii}}{\Pi_i P_i}\right)^{1-\sigma} \tag{3}$$

In Eq (3),  $\tau_{ii}$  represents the domestic trade costs of country *i*. By rewriting Eq (3), we can obtain the product of the outward multilateral resistance and the inward multilateral resistance of country *i*:

$$\Pi_i P_i = t_{ii} \left( \frac{X_{ii}/Y_i}{Y_i/Y^W} \right)^{\frac{1}{(\sigma-1)}}$$
(4)

Similarly, the product of the outward multilateral resistance and the inward multilateral resistance of country *j* can be obtained:

$$\Pi_j P_j = t_{jj} \left( \frac{X_{jj} / Y_j}{Y_j / Y^W} \right)^{\frac{1}{(\sigma - 1)}}$$
(5)

We multiply the two sides of Eq (1) and Eq (2) to arrive at a two-way gravity equation:

$$X_{ij}X_{ji} = \left(\frac{Y_iY_j}{Y^W}\right)^2 \left(\frac{t_{ij}t_{ji}}{\Pi_i P_i \Pi_j P_j}\right)^{1-\sigma}$$
(6)

We substitute Eqs (4) and (5) into Eq (6) to get:

$$\left(\frac{t_{ij}t_{ji}}{\mathsf{t}_{ii}t_{jj}}\right) = \left(\frac{X_{ii}X_{jj}}{X_{ij}X_{ji}}\right)^{\frac{1}{(\sigma-1)}} \tag{7}$$

Considering the asymmetry of trade costs between country *i* and country  $j(t_{ij} \neq t_{ji})$ , their domestic trade costs are also different  $(t_{ii} \neq t_{ji})$ . Following Novy's approach, we get the equation of trade costs in tariff equivalent by calculating the geometric mean of trade costs in all directions and then subtracting one from both sides in Eq (7). It should be noted that the derivation of a micro-founded measure of trade costs has been done by Novy, Dennis (see Novy [24] for the details of the discussion). This approach is similar to the models proposed by Head and Ries [66] and Head and Mayer [69].

$$\tau_{ij} = \left(\frac{t_{ij}t_{ji}}{t_{ii}t_{jj}}\right)^{\frac{1}{2}} - 1 = \left(\frac{X_{ii}X_{jj}}{X_{ij}X_{ji}}\right)^{\frac{1}{2(\sigma-1)}} - 1$$
(8)

In Eq (8),  $\tau_{ij}$  denotes the tariff equivalent trade costs, which measure the bilateral trade costs of the two trading partners  $(t_{ij}t_{ji})$  relative to their respective domestic trade costs  $(t_{ii}t_{jj})$ . Assuming all other variables remain constant, Eq (8) illustrates that if the bilateral trade  $(X_{ij}X_{ji})$  of the two trading partners grows faster than their respective domestic trade  $(X_{ii}X_{jj})$ , trade costs will decrease.

**4.1.2 Measurement method of average trade costs.** This portion refers to the method derived by Arvis et al. [27] to compute the annual average trade costs between Cambodia and all of its trading partners. The geometric average of the actual total trade between country *i* and its trading partner *j* can be defined as

$$\bar{X}_{ij} = \bar{X}_{ji} = (X_{ij} \times X_{ji})^{\frac{1}{2}}.$$
 (9)

The authors set the following equation, based on the bilateral trade cost model of Novy (2013), to get the formula for calculating average trade costs ( $\bar{\tau}_t$ ):

$$\bar{\tau}_{t} = \left(\frac{\sum_{j \neq i} \bar{X}_{ij}}{(X_{ii})^{\frac{1}{2}} \sum_{j \neq i} (X_{jj})^{\frac{1}{2}}}\right)^{\frac{1}{1-\sigma}} - 1$$
(10)

In Eq (10), the annual average trade costs can be measured using the domestic trade data of both countries, the elasticity of substitution between goods, and the geometric average of the actual total trade as defined in Eq (9).  $\bar{\tau}_i$  represents the annual average trade costs of Cambodia with all of its trading partners in the given year *t* (*t* = 1993, 1994,..., 2019).

#### 4.2 Decomposition model of trade growth

This sub-section presents an overview of the existing literature concerning the factors that contribute to the expansion of trade and introduces the decomposition model of trade growth. Krugman [70] and Feenstra [71] argue that one of the significant drivers of trade expansion is the decline in trade costs. Baier and Bergstrand [62] investigated the factors influencing bilateral trade in 16 OECD countries between 1958 and 1988. Their findings indicated that income, tariffs, and transportation costs had significant impacts on trade growth, with income growth having the largest impact (67%), followed by a reduction in tariffs (25%) and a decrease in transportation costs (8%). Jacks et al. [21] decomposed the growth of global trade based on two main determinants: income and trade costs. They found that the growth of global trade between 1870 and 1913 was driven by 44 percent due to a reduction in trade costs. Jacks et al. [4] discovered that the decline in trade costs accounted for almost 31 percent of the global trade growth between 1950 and 2000. In summary, previous studies have confirmed that reducing trade costs is necessary for expanding international trade.

So far, there has been a lack of research on the determinants of trade growth in Cambodia. Hence, the question arises: has the decline in trade costs played a role in Cambodia's trade growth over the past two decades? In order to answer this question, this section decomposes the trade growth equation derived by Novy [24] into three components: income, trade costs, and multilateral resistance.

First, we take the natural logarithm for both sides of Eq (6). Then, we derive the first-order difference of the natural logarithm result to obtain Eq (11):

$$\Delta \ln(X_{ij}X_{ji}) = 2\Delta \ln\left(\frac{Y_iY_j}{Y^W}\right) + (1-\sigma)\Delta \ln(t_{ij}t_{ji}) - (1-\sigma)\Delta \ln(\Pi_iP_i\Pi_jP_j)$$
(11)

Eq (11) indicates that the bilateral trade growth,  $\Delta \ln(X_{ij}X_{ji})$ , is influenced by three factors: the first is the change in each country's income compared to the world's income,  $\Delta \ln \left(\frac{Y_i Y_j}{Y^W}\right)$ ; the second is the change in the bilateral trade costs,  $\Delta \ln(t_{ij}t_{ji})$ ; and the third is the change in the level of multilateral resistance between trading partners,  $\ln(\Pi_i P_i \Pi_j P_j)$ . Since the value of the bilateral trade costs  $t_{ij}$  and  $t_{ji}$  are unknown, we substitute Eq (8) into Eq (11) to obtain the following Eq (12):

$$\Delta \ln(X_{ij}X_{ji}) = 2\Delta \ln\left(\frac{Y_iY_j}{Y^W}\right) + 2(1-\sigma)\Delta \ln(1+\tau_{ij}) - 2(1-\sigma)\Delta \ln(\Phi_i\Phi_j)$$
(12)

Among them,  $\Phi_i = \left(\frac{\Pi_i P_i}{t_{ii}}\right)^{1/2}$  and  $\Phi_j = \left(\frac{\Pi_j P_j}{t_{jj}}\right)^{1/2}$ 

In Eq (12),  $\Phi_i$  and  $\Phi_j$  denote the multilateral resistance of country *i* and country *j*, respectively, relative to their domestic trade costs.

Further, we divide both sides of Eq (12) by  $\Delta \ln(X_{ij}X_{ji})$  to obtain the bilateral trade decomposition equation as follows:

Fig 2 explains that the growth of bilateral trade between the two trading partners comes from the contribution of three determinants: (A) the growth of income, (B) the reduction in bilateral trade costs, and (C) the decrease in multilateral resistance. Assuming bilateral trade barriers were constant over time between both countries, the combined contribution of (B) and (C) would amount to zero. As a result, the growth of income (A) would be the sole contributor to trade growth. If bilateral trade costs decrease (i.e.,  $\Delta \ln(1+\tau_{ij})<0$ ), it indicates that the contribution of the decline in bilateral trade costs to the growth of bilateral trade is positive, inferring that (B) is positive. If the multilateral resistance declines (i.e.,  $\Delta \ln(\Phi_i \Phi_j)<0$ ), then (C) becomes negative. This negative impact can be explained as the trade diversion effect due to a

$$100\% = \frac{2\Delta \ln\left(\frac{Y_i Y_j}{Y^W}\right)}{\Delta \ln(X_{ij} X_{ji})} + \frac{2(1-\sigma)\Delta \ln(1+\tau_{ij})}{\Delta \ln(X_{ij} X_{ji})} - \frac{2(1-\sigma)\Delta \ln(\Phi_i \Phi_j)}{\Delta \ln(X_{ij} X_{ji})}$$
(13)

Fig 2. Eq 13.

https://doi.org/10.1371/journal.pone.0311754.g002

decline in multilateral resistance. It can also be interpreted that when barriers to international trade decrease with other countries, there is a subsequent increase in trade between those countries, resulting in a decline in bilateral trade between countries *i* and *j*.

To solve  $2(1-\sigma)\Delta \ln(1+\tau_{ij})$ , we recall Eq (8).

$$2(1-\sigma)\Delta\ln(1+\tau_{ij}) = \Delta\ln(X_{ij}X_{ji}) - \Delta\ln(X_{ii}X_{jj})$$
(14)

To solve  $2(1-\sigma)\Delta \ln(\Phi_i \Phi_j)$ , we recall Eqs (4) and (5).

$$2(1-\sigma)\Delta\ln(\Phi_i\Phi_j) = \Delta\ln\left(\frac{Y_i/Y^W}{X_{ii}/Y_i}\right) + \Delta\ln\left(\frac{Y_j/Y^W}{X_{jj}/Y_j}\right)$$
(15)

The decomposition equation exemplifies that the growth of bilateral trade does not depend on the value of the elasticity of substitution between goods ( $\sigma$ ).

By substituting Eqs (14) and (15) into Fig 2, we can obtain the following Eq (16):

$$100\% = \frac{2\Delta \ln\left(\frac{Y_i/Y_j}{Y^W}\right)}{\Delta \ln(X_{ij}X_{ji})} + \frac{\Delta \ln(X_{ij}X_{ji}) - \Delta \ln(X_{ii}X_{jj})}{\Delta \ln(X_{ij}X_{ji})} - \frac{\Delta \ln\left(\frac{Y_i/Y^W}{X_{ii}/Y_i}\right) + \Delta \ln\left(\frac{Y_j/Y^W}{X_{ij}/Y_j}\right)}{\Delta \ln(X_{ij}X_{ji})}$$
(16)

#### 4.3 Sample, data sources, and parameter settings

**4.3.1 Sample.** In this study, the top 30 trading partners are selected based on the availability of data and their stable bilateral trade volumes with Cambodia over the past two decades. In 2019, all these trading partners collectively made up approximately 94 percent of Cambodia's total exports and imports, based on the data from the International Monetary Fund (IMF)'s Direction of Trade Statistics (DOTS) [72]. As of 2019, Cambodia and its top 30 trading partners contributed to over 76 percent of the world's total exports and imports, 73 percent of the world's economy measured by the purchasing power parity (PPP), and more than half of the global population [72, 73]. Therefore, this representative sample can be used to infer Cambodia's overall trade pattern with the world. These top 30 trading partners include 13 countries in the G20 (Australia, Canada, China, Germany, France, India, Indonesia, Italy, Japan, the Republic of Korea, the Russian Federation, the United Kingdom, and the United States), eight countries in the European Union (EU) that are not the members of the G20 (Austria, Belgium, Denmark, Finland, Ireland, the Netherlands, Spain, and Sweden), five countries in the RCEP, that are not the members of the G20 (Malaysia, New Zealand, Singapore, Thailand, and Vietnam), and other major trading partners (Hong Kong (China), Norway, Switzerland, and the Taiwan Province of China). This paper uses the annual data for the 27-year period spanning from 1993 to 2019, equivalent to a total of 810 observations ( $N = 30 \times T = 27$ ). It is important to note that we select the post-1993 sample to highlight the development of Cambodia's trade liberalization, as the country has only started official trade with many countries and regions since implementing a market-oriented economy in 1993.

**4.3.2 Data sources.** The primary source of the bilateral export data in free-on-board (FOB) is derived from the IMF's Direction of Trade Statistics (DOTS) database. However, the data of some trading partners are not available from the IMF's DOTS database. Therefore, where possible, the missing data are supplemented using the data from the OECD's Structural Analysis (STAN) database and the United Nations Conference on Trade and Development (UNCTAD) database (refer to the <u>S1 Appendix</u> for the detailed descriptions of this procedure). The domestic trade data are based on the latest update (second edition) dataset from the International Trade and Production Database for Estimation Release 2 (ITPD-E-R02), published in

Variable	Description	Sources
$X_{ij}$	Nominal export from Cambodia to country $j$ (Thousand USD)	IMF's DOTS, OECD's STAN and UNCTAD
$X_{\rm ji}$	Nominal export from country <i>j</i> to Cambodia (Thousand USD)	
$X_{ii}$	Domestic trade of Cambodia (Thousand USD)	ITPD-E-R02
$X_{jj}$	Domestic trade of country $j$ (Thousand USD)	
$Y_{i}$	Nominal income of Cambodia (Thousand USD)	IMF's WEO
$Y_j$	Nominal income of country <i>j</i> (Thousand USD)	
$Y^W$	World's nominal income (Thousand USD)	

Table 1. Description of the variables and data sources.

Source: Authors' Elaboration

https://doi.org/10.1371/journal.pone.0311754.t001

July 2022. The ITPD-E-R02 database covers the data for 265 countries or regions and 170 industries in the four sectors, including agriculture, mining & energy, manufacturing, and services, from 1986 to 2019 [74, 75]. It is important to emphasize that the scope of this research is limited to trade in the goods sector and total trade costs. Due to the unavailability of data on domestic trade in other sectors for Cambodia and other developing countries, this research does not analyze disaggregated trade costs by industries. To obtain the total domestic trade data in the goods sector, we aggregate the three industries, including agriculture, mining & energy, and manufacturing, from the ITPD-E-R02 database. The GDP data are also needed to decompose the growth of the Cambodian trade using Eq (16), and they are taken from the IMF's World Economic Outlook (WEO) database published in October 2022. In order to capture the price indices in what Anderson and van Wincoop [58] refer to as "multilateral resistance", trade volumes and income are estimated using the nominal value. All the data are measured in thousands of current US dollars for the corresponding year. Table 1 summarizes the variables used to measure trade costs, their descriptions, and their data sources, and Table 2 presents the descriptive statistics of these variables.

**4.3.3 Parameter settings.** To calculate trade costs using Eqs (8) and (10), it is necessary to make the parameter assumption. Specifically, determining the exact value of the elasticity of substitution across goods is challenging. Based on a summary of estimated results from various existing literature, as reported by Anderson and van Wincoop [10], the elasticity of substitution across goods ( $\sigma$ ) is likely to fall between 5 and 10. Novy [24] demonstrates that changes in trade costs are not affected by changes in the elasticity of substitution. He observes that between 1970 and 2000, when  $\sigma = 10$ , the average bilateral trade costs of the United States

	1					
Variable	Observations	Mean value	Standard deviation	Minimum value	Maximum value	
X <sub>ij</sub>	807	148244.6	371330	0.1113	4414284	
X <sub>ji</sub>	807	277875.2	830814.8	7.153	8000644	
X <sub>ii</sub>	810	3009412	2335690	618366.1	7859559	
X <sub>ii</sub>	810	4.97E+08	1.42E+09	5257	1.61E+10	
Y <sub>i</sub>	810	9968074	7276862	2427000	2.71E+07	
Y <sub>i</sub>	810	1.48E+09	2.82E+09	1.67E+07	2.14E+10	
$Y^W$	810	5.41E+10	2.07E+10	2.61E+10	8.77E+10	

Table 2. Descriptive statistics of the variables.

#### Source: Authors' Calculation

https://doi.org/10.1371/journal.pone.0311754.t002

decreased from 54 percent to 31 percent, a decline of 42 percent; when  $\sigma$  = 8, trade costs decreased from 74 percent to 42 percent, a decline of 44 percent. Therefore, in this paper, we set the value of the elasticity of substitution across products ( $\sigma$ ) to 8, as suggested by the previous studies [10, 24, 27].

# 5. Empirical results and discussions

# 5.1 Measurement results and analyses of bilateral trade costs between Cambodia and major trading partners

In this section, we present the results of the ad valorem equivalent bilateral trade costs between Cambodia and 30 of its top trading partners between 1993 and 2019.

Table 3 presents the levels and the percentage changes in the bilateral trade costs (ad valorem equivalent) between Cambodia and 30 of its top trading partners between 1993 and 2019. The trading partners are organized in ascending order based on their relative bilateral trade costs with Cambodia as of 2019 across regions. There are significant variations in the absolute value of the bilateral trade costs across trading partners. For instance, Cambodia's bilateral trade costs with Hong Kong (China) were as low as 1.35% in 2019, while this number reached 251.25% with Norway in the same year. Notably, Norway is the only trading partner with trade costs exceeding 200%. Over the years from 1993 to 2019, trade costs of Hong Kong (China), Thailand, Vietnam, and Singapore remained below 100%, which were consistently lower than those of other trading partners. We offer a few possible explanations for this phenomenon. First, Hong Kong (China) and Singapore function as the major entrepot hubs in the Asia-Pacific region, facilitating substantial re-export trade. This means that their domestic trade flows are relatively smaller than their foreign trade flows, leading to lower trade costs. Second, Hong Kong (China) and Singapore boast some of the lowest tariffs and exhibit a high degree of trade openness. Third, both trading partners are geographically close to Cambodia, resulting in low transportation costs. Most importantly, Singapore has shared the Free Trade Agreement (FTA) with Cambodia under the ASEAN Free Trade Area (AFTA) since 1999, which is vital for reducing trade barriers between the two countries. Refer to the following section for a comprehensive discussion of the bilateral trade costs between Cambodia and its neighboring countries (Thailand and Vietnam).

In terms of the regional analysis, except for Indonesia in Southeast Asia, all trading partners whose trade costs were over 100% in 2019 are situated in South Asia, Oceania, Europe, and North America, with the majority being in Europe. In general, the trade costs of the short-distance trading partners in Southeast Asia and East Asia are lower than most of the long-distance counterparts in Europe and North America. This finding infers that geographic distance is one of the most critical factors affecting trade costs, which aligns with the theoretical framework of the gravity model. The hypothesis of this model suggests that trading partners located far from each other tend to engage in less trade due to high transportation costs [58, 60].

Another notable finding is that despite being a part of the ASEAN Free Trade Area (AFTA) with Indonesia since 1999, the ASEAN-Australia-New Zealand Free Trade Area (AANZFTA) with Australia and New Zealand since 2010, and the ASEAN-India Free Trade Area (AIFTA) with India since 2010, Cambodia's bilateral trade costs with these countries have remained relatively high during the specific time period. Therefore, it is essential to focus on ongoing efforts to reduce trade barriers and improve trade facilitation, particularly in the context of the recent Regional Comprehensive Economic Partnership (RCEP) framework, which is already in force on January 1, 2022, involving countries such as Australia, Indonesia, and New Zealand.

Region	Trading Partner	$ au_{1993}$	$ au_{2019}$	Δ%
Southeast Asia	Thailand	50.12	33.60	-32.96
	Vietnam	78.81	40.59	-48.49
	Singapore	67.78	54.11	-20.16
	Malaysia	96.18	86.08	-10.50
	Indonesia	145.43	114.14	-21.51
East Asia	Hong Kong (China)*	53.52	1.35	-97.48
	Republic of Korea**	201	56.82	-71.73
	Japan	96.82	59.12	-38.95
	China	197.06	62.22	-68.43
	Taiwan Province of China	173.99	76.53	-56.01
South Asia	India	264.06	167.32	-36.63
Oceania	Australia	237.57	151.40	-36.27
	New Zealand	308.38	167.00	-45.85
North America	United States	196.59	98.70	-49.79
	Canada	245.68	110.37	-55.07
Europe	United Kingdom	255.07	66.60	-73.89
	Switzerland	234.47	90.94	-61.21
	Belgium	191.18	94.79	-50.42
	Denmark	272.91	98.48	-63.91
	Netherlands	130.32	110.30	-15.36
	France	225.63	117.00	-48.15
	Germany	147.90	121.92	-17.57
	Italy	261.90	129.26	-50.65
	Spain	433.46	138.05	-68.15
	Ireland	431.88	144.74	-66.49
	Austria	259.83	171.24	-34.09
	Sweden	350.54	171.26	-51.14
	Finland	198.40	172.69	-12.96
	Russian Federation	189.14	196.23	+3.74
	Norway	331.90	251.25	-24.30

#### Table 3. Bilateral trade costs (tariff equivalent) between Cambodia and its 30 major trading partners between 1993 and 2019 ( $\sigma = 8$ ).

Notes: All values are in percent and recorded with two decimal places—source: Authors' calculation using Eq (8)

\* The bilateral trade costs between Cambodia and Hong Kong (China) have shown a negative value in some years. Therefore, the data for the year 2016 are used as the end-year data

\*\* For the Republic of Korea, trade costs are measured from the year 1995 due to the unavailability of data before the year 1995.

https://doi.org/10.1371/journal.pone.0311754.t003

Regarding the magnitudes of the changes, there have been significant shifts in the bilateral trade costs between Cambodia and its major trading partners. All trading partners have reduced their respective trade costs, except for the Russian Federation. It is clear from the findings that there is an opportunity for further trade optimization between Cambodia and the Russian Federation. Cambodia's bilateral trade costs with Hong Kong (China), the United Kingdom, the Republic of Korea, China, Spain, Ireland, Denmark, and Switzerland experienced a remarkable decline, exceeding 60% between 1993 and 2019. It is important to note that in 1993, Cambodia's bilateral trade costs with Spain and Ireland were extremely high, reaching 433.46% and 431.88%, respectively. However, by 2019, these figures dramatically decreased to 138.05% and 144.74%, showing a decline of over 65%. In contrast, Germany (– 17.57%), the Netherlands (–15.36%), Finland (–12.96%), and Malaysia (–10.50%) had the lowest decline in trade costs.

Trading partner	$ au_{1993}$	$ au_{2019}$	Δ%
China	197.06	62.22	-68.43
United States	196.59	98.70	-49.79
Thailand	50.12	33.60	-32.96
Vietnam	78.81	40.59	-48.49
Japan	96.82	59.12	-38.95
Germany	147.90	121.92	-17.57
United Kingdom	255.07	66.60	-73.89
Canada	245.68	110.37	-55.07
Republic of Korea*	201	56.82	-71.73
Singapore	67.78	54.11	-20.16

Table 4. Bilateral trade costs (tariff equivalent) between Cambodia and its ten largest trading partners between 1993 and 2019 ( $\sigma = 8$ ).

Notes: All values are in percent and recorded with two decimal places—source: Authors' calculation using Eq (8). \* For the Republic of Korea, trade costs are measured from the year 1995 due to the unavailability of data before the year 1995.

#### https://doi.org/10.1371/journal.pone.0311754.t004

Table 4 displays the levels and the percentage reductions in Cambodia's relative bilateral trade costs with its ten largest trading partners between 1993 and 2019. The top ten largest trading partners based on the 2019 share of Cambodia's total trade volumes listed in descending order are China (23.93%), the United States (13.17%), Thailand (10.42%), Vietnam (8.58%), Japan (5.64%), Germany (3.35%), the United Kingdom (2.96%), Canada (2.67%), the Republic of Korea (2.47%), and Singapore (2.25%), according to the data from the IMF's DOTS database [72]. Concerning the percentage declines, only the United Kingdom, the Republic of Korea, and China reduced their bilateral trade costs with Cambodia by more than two-thirds between 1993 and 2019. In contrast, the lowest declines in trade costs were recorded in Thailand (-32.96%), Singapore (-20.16%), and Germany (-17.57%).

Regarding its neighboring countries, Cambodia's bilateral trade costs were 33.6% with Thailand and 40.59% with Vietnam in 2019. These figures show that Thailand and Vietnam have the second and third lowest trade costs, respectively, following Hong Kong (China) (see Table 3). In 2019, Thailand and Vietnam were Cambodia's third and fourth-largest trading partners, respectively, behind China and the United States. Notably, Thailand and Vietnam are the only two of Cambodia's top 10 trading partners with trade costs below 50%. There are several reasons for low trade costs with these countries. First, Cambodia's geographic advantage as a neighboring country of Thailand and Vietnam allows it to enjoy lower transportation costs when trading with these countries. Next, Cambodia and its neighbors have developed close economic ties due to their long-standing shared history and cultural similarities. Finally, they have been a part of the ASEAN Free Trade Area (AFTA) since 1999, which is significant for reducing trade barriers among its members. This finding aligns with the literature showing that trade integration has rapidly accelerated across geographically close countries. For example, Novy [24] found that in the year 1993, the bilateral trade costs (tariff equivalent) between the two neighbors (the United States and Canada) were 31%, which is significantly lower than trade costs between Cambodia and Thailand (50.12%) and Cambodia and Vietnam (78.81%) in the same year. These numbers show that in 1993, Cambodia's bilateral trade costs with Thailand and Vietnam were 61.68% and 154.23%, respectively, higher than those between the United States and Canada. In 2019, the bilateral trade costs between Cambodia and Thailand stood at 33.60%. This figure is comparable to 31% of the United States and Canada bilateral trade costs in 1993 and 33% of the United States and Mexico bilateral trade costs in 2000, as

Year	1993	1994	1995	1996	1997	1998	1999	2000	2001
Trade Costs	139.7	141.76	144.46	150.45	140	136.51	139	132.82	133.18
Year	2002	2003	2004	2005	2006	2007	2008	2009	2010
Trade Costs	127.57	130.16	126.97	131.87	131.29	131.97	132.17	124.22	118.94
Year	2011	2012	2013	2014	2015	2016	2017	2018	2019
Trade Costs	120.09	117.38	111.57	119.73	109.97	103.27	99.28	96.55	90.21

Table 5. Cambodia's average trade costs from 1993 to 2019 ( $\sigma = 8$ ).

Notes: All values are in percent and recorded with two decimal places—source: Authors' calculation using Eq (10).

https://doi.org/10.1371/journal.pone.0311754.t005

reported in Novy's survey [24]. Although there have been significant declines in the bilateral trade costs between Cambodia and its neighboring countries over the past two decades, these costs have remained higher than those observed between highly developed countries nearly 30 years ago. This comparison highlights the significant progress made by highly developed countries in reducing trade costs over time.

Another noteworthy finding is that Germany and Canada, two of Cambodia's top ten trading partners, had trade costs exceeding 100% in 2019 (refer to <u>Table 4</u>). Simultaneously, the bilateral trade costs between Cambodia and the United States were also relatively high, approaching 100%. These three countries' trade costs were higher than the average of 90.21% in 2019 (see <u>Table 5</u>).

#### 5.2 Measurement results and analyses of Cambodia's average trade costs

For a comprehensive understanding of Cambodia's trade costs and their drastic changes over time, it is crucial to measure the country's average trade costs with all of its trading partners. However, due to the limited data availability, this study includes only 30 of Cambodia's major trading partners. In this sub-section, we compute Cambodia's average trade costs using a model-based equation. This method is considered superior to the simple or trade-weighted average, as it does not give much weight to large trading partners [27]. After calculating the bilateral trade costs between Cambodia and each trading partner using Eq (8), we compute Cambodia's average trade costs based on Eq (10), and the estimated results are shown in Table 5.

Table 5 shows that Cambodia's average trade costs (ad valorem equivalent) peaked at 150.45% in 1996. We can observe that before the country joined the ASEAN in 1999, its average trade costs remained high, exceeding 136% over the years from 1993 to 1999. Following Cambodia's accession to the ASEAN, there has been a notable reduction in its average trade costs. Specifically, its average trade costs decreased from 139% in 1999 to 132.82% in 2000, equivalent to a decline of 4.44%. Despite becoming a member of the WTO in 2004, Cambodia had experienced persistently high and volatile trade costs until 2014. Since then, these costs have gradually decreased, falling below 100% in 2017 and reaching 90.21% in 2019.

As displayed in Fig 3, Cambodia's average trade costs have shown a clear downward trend since 2014. Between 1993 and 2019, Cambodia's average trade costs decreased by 35.43%. These costs fell by a huge margin of 24.66% between 2014 and 2019 alone, adding to a considerable decline in the overall trade costs between 1993 and 2019. These findings suggest that Cambodia has committed to fulfilling its trade obligations under the WTO framework and has consistently increased its trade openness to global markets.

In order to gain insights into the scales and the evolutions of Cambodia's trade costs over the past two decades, it is crucial to conduct a comparative analysis of its average trade costs with other countries, as documented in the literature that used the same measuring methods.



https://doi.org/10.1371/journal.pone.0311754.g003

In the year 2000, Cambodia's average trade costs (tariff equivalent) stood at 132.82% (see Table 5). This number is 41.29% higher than the average trade costs of 13 OECD countries, which were recorded at 94% in the same year, as reported by Novy [24]. In 2004, the average trade costs of 37 major developed and industrializing countries were only 66%, according to a study conducted by Milner and Mcgowan [31]. This figure is almost half as low as the average trade costs of Cambodia, which were nearly 127% in the same year. In 2019, Cambodia's average trade costs were approximately 90%. It is evident that although there has been a downward trend in Cambodia's average trade costs in recent years, they remain 36% higher than those in major developed and industrializing countries 16 years ago.

As shown in Fig 3, Cambodia's average trade costs have declined more rapidly since 2014. We provide a few possible explanations for this phenomenon. First, most of Cambodia's top trading partners recovered from major crises, including the 2008–2009 global financial crisis, the European debt crisis in 2010, and the European Union recession in 2013. Therefore, various countries and regions relaxed trade protectionism, which led to a decline in trade costs between Cambodia and its major trading partners. Second, trade costs between Cambodia and its major trading partners. Second, trade costs between Cambodia and its major trading partners. Second, trade costs between Cambodia and its major trading partners. Second, trade costs between Cambodia's average trade costs. This can be attributed to infrastructure development, tariff rate adjustments, and other measures implemented as part of the BRI, launched by President Xi Jinping in September 2013. Thus, to delve deeper into the impact of the BRI on Cambodia's trade costs, as discussed in the next section, we classify 30 trading partners into two groups: the BRI trading partners and the non-BRI trading partners.

Finally, the rapid decline in average trade costs after 2014 coincided with Cambodia's transformation from a low-income to a lower-middle-income country in 2015, according to the World Bank's country classifications by income level [41]. We can observe that Cambodia experienced consistently high trade costs throughout its status as a low-income country from 1993 to 2014. This finding aligns with the literature indicating that trade costs decrease as the country's per capita income increases. Low-income countries have not witnessed any significant improvement in their trade costs compared to other income groups, and those countries continue to face difficulties in reducing trade barriers [27]. Therefore, to provide a more indepth understanding of the dynamic changes in Cambodia's average trade costs with major trading partners based on the different levels of their economic development, we classify 30 trading partners into two groups: developed economies and developing & emerging economies. With the same measure stated in Eq (10), Cambodia's average trade costs with its trading partners under different classifications can be determined, and the results are elaborated in the following sections.

**5.2.1 A comparison between Cambodia's trade costs with developing economies and developed economies.** To gain a comprehensive understanding of various degrees of trade costs among trading partners based on their respective levels of economic development, we use the economy groupings classified by the International Monetary Fund (IMF) to divide the whole sample into two groups: developed economies and developing & emerging economies [76]. Seven developing and emerging economies include China, India, Indonesia, Malaysia, the Russian Federation, Thailand, and Vietnam. Twenty-three developed economies include Australia, Austria, Belgium, Canada, Denmark, Finland, France, Germany, Hong Kong (China), Ireland, Italy, Japan, the Republic of Korea, the Netherlands, New Zealand, Norway, Singapore, Spain, Sweden, Switzerland, the Taiwan Province of China, the United Kingdom, and the United States.

In terms of the scales, it can be seen from Fig 4 that the average trade costs between Cambodia and developing and emerging economies were consistently lower than those of the developed economies from 1993 to 2006. From 2007 to 2012, average trade costs between Cambodia and the two groups were similar. However, Cambodia's average trade costs with developing and emerging economies experienced a slight rebound in 2009 and 2010, surpassing the average trade costs of developed economies by 11.25% and 3%, respectively. Between



Fig 4. Changes in Cambodia's average trade costs with developed economies and developing & emerging economies from 1993 to 2019 ( $\sigma = 8$ ). Source: Authors' calculation using Eq (10).

https://doi.org/10.1371/journal.pone.0311754.g004

2013 and 2014, there was a marginal rise in average trade costs between Cambodia and the two groups. One potential explanation for this phenomenon might be attributed to the 2008–2009 global financial crisis and the subsequent European Union recession in 2013. These economic downturns contributed to a rise in trade protectionism across different countries and regions, causing an increase in trade costs. These findings are consistent with the literature, suggesting that an uncertainty shock during a crisis directly hinders cross-border trade flows while simultaneously increasing average trade costs for all trading partners [77].

Contrary to what one would expect, Cambodia's average trade costs with developing and emerging economies are lower than those with developed economies when engaging in trade. This finding also aligns with the substantial contribution of developing and emerging economies to Cambodia's total trade volumes. For instance, seven emerging and developing economies, namely China, India, Indonesia, Malaysia, the Russian Federation, Thailand, and Vietnam, made up almost 48% of Cambodia's overall trade in 2019. In contrast, the remaining twenty-three developed economies accounted for only 46.4% of the country's total trade. Due to its proximity, Cambodia has benefited from lower transportation costs when trading with developing and emerging economies like Thailand, Vietnam, China, India, Indonesia, and Malaysia. Therefore, their trade costs are less likely to be affected by their level of per capita income.

Regarding the magnitudes of the changes, the average trade costs of developed economies showed a downward trend after reaching a peak at 168.57% in 1995. These costs decreased by 36.17%, from 157.61% in 1993 to 100.60% in 2019. In contrast, the average trade costs of developing and emerging economies increased from 106.29% in 1993 to a peak of 132.67% in 2007. Between 1993 and 2019, average trade costs between Cambodia and developing and emerging economies decreased by 29.42%, from 106.29% in 1993 to 75% in 2019. The findings show that since 2014, there has been a notable decline in the trend of average trade costs between Cambodia and both groups. Cambodia's average trade costs with developing and emerging economies decreased by 35.17%, from 115.71% in 2014 to 75% in 2019, equivalent to 8.3% per year. This figure surpasses the 17.85% decline in developed economies between 2014 and 2019, where average trade costs dropped from 122.46% in 2014 to 100.60% in 2019, corresponding to 3.86% annually. The decline in trade costs is likely due to developing and emerging countries accelerating the liberalization of their trade policies. In addition, contributing to this reduction in trade costs is also influenced by Cambodia's trade openness and economic integration with developing and emerging countries through various regional trade agreements (RTAs) under the framework of the ASEAN Free Trade Area (AFTA effective from 1999), the ASEAN-China Free Trade Area (ACFTA effective from 2005), and the ASEAN-India Free Trade Area (AIFTA effective from 2010). In this analysis, all trading partners categorized as developing and emerging economies comprise lower-middle-income countries (India and Vietnam) and upper-middle-income countries (China, Indonesia, Malaysia, the Russian Federation, and Thailand) according to the World Bank country classifications by income levels in 2019 [78]. Therefore, this finding is consistent with the existing literature indicating that developing countries in the upper-middle-income bracket have achieved a more accelerated reduction in trade costs compared to developed countries [27]. It also reflects that trade frictions remain despite long-standing trade ties between Cambodia and developed economies. For example, in May 2015, Cambodian bicycle exports to the European Union market were subject to a 48.5% import duty under the rule of origin [79].

**5.2.2 A comparison between Cambodia's trade costs with the Belt and Road Initiative** (**BRI**) and the non-BRI trading partners. In light of the growing interest in the Belt and Road Initiative (BRI) among academics and policymakers in recent years, we also consider the impact of the BRI on Cambodia's trade costs. The year 2023 marked the tenth anniversary of

the BRI since it was introduced by President Xi Jinping in September 2013. By the end of August 2023, more than 150 countries and over 30 international organizations have participated and signed cooperation agreements with China under the BRI framework [80]. Cambodia was among the first countries to support the BRI projects, and the Royal Government of Cambodia (RGC) signed the Memorandum of Understanding (MoU) with China in 2016 under the BRI's framework to develop the infrastructure [81, 82]. The existing literature on trade, investment, and infrastructure development related to the BRI has demonstrated that countries or regions adjacent to China have experienced the extensive scales of development brought about by this initiative [36, 83, 84]. Hence, there could be an expansion of trade between Cambodia and its trading partners along the BRI corridors despite Cambodia not being a signatory to the BRI with other countries besides China. We consult several sources to determine the major trading partners along the BRI corridors. First, the initial list of 64 countries along the BRI corridors and China was published by the Chinese government in 2015, including Cambodia, China, India, Indonesia, Malaysia, the Russian Federation, Singapore, Thailand, and Vietnam [82]. We also include Hong Kong (China), Japan, the Republic of Korea, and the Taiwan Province of China as Cambodia's trading partners along the BRI corridors, according to the lists identified by the OECD and the World Bank [36, 85]. Thus, we finally classify 12 major trading partners of Cambodia along the BRI corridors for our analysis, which comprise China, Hong Kong (China), India, Indonesia, Japan, the Republic of Korea, Malaysia, the Russian Federation, Singapore, the Taiwan Province of China, Thailand, and Vietnam. We also categorize the other 18 major trading partners of Cambodia as the non-BRI countries, which include Australia, Austria, Belgium, Canada, Denmark, Finland, France, Germany, Ireland, Italy, the Netherlands, New Zealand, Norway, Spain, Sweden, Switzerland, the United Kingdom, and the United States.

Fig 5 indicates that average trade costs between Cambodia and its trading partners along the BRI corridors decreased by nearly 34.78%, from 108.16% in 2014 to 70.54% in 2019, equivalent to 8.2% per year. This number is significantly higher than the 16.14% decline in the non-BRI trading partners between 2014 and 2019, with the average trade costs falling from 136.34% in 2014 to 114.34% in 2019, corresponding to 3.46% per year. Overall, average trade costs between Cambodia and the trading partners along the BRI corridors have decreased by more than twice as fast as the non-BRI trading partners since 2014. A possible explanation for the rapid decline in trade costs between Cambodia and its trading partners along the BRI corridors may also reflect better infrastructure development and improved trade facilitation. Furthermore, the large-scale BRI development projects are more likely to have significant impacts on the markets of infrastructure-construction-related products, such as construction equipment, cement, iron, and steel, among others, between Cambodia and its trading partners along the BRI corridors, especially those along the China-Indochina Peninsula Economic Corridor. This finding is consistent with the literature. For example, a study conducted by de Soyres et al. [36] demonstrates that economies located along the BRI corridors, where transportation infrastructure projects have been developed, witness significant benefits. These benefits include a notable reduction in shipment time of up to 11.9% and a decrease in trade costs by up to 10.2%. In comparison, the global average decline in shipment time ranges from 1.2% to 2.5%, leading to a decrease in the global aggregate trade costs between 1.1% and 2.2%.

#### 5.3 Decomposing the growth of Cambodia's trade

Cambodia's adoption of a market-based economy in the early 1990s, followed by its membership in the ASEAN in 1999 and the WTO in 2004, has contributed significantly to the rapid growth of its trade. From 1993 to 2019, Cambodia's total trade volumes increased 46.5 times,





https://doi.org/10.1371/journal.pone.0311754.g005

from US\$ 755 million to US\$ 35.1 billion. Therefore, a couple of questions arise: What factors have led to the growth of Cambodia's trade? How do trade costs contribute to trade expansion? This section decomposes Cambodia's trade growth between 1993 and 2019 into three components: income, trade costs, and multilateral resistance. This portion helps us understand the determinants that have driven the growth of Cambodia's trade over the past two decades.

**5.3.1 Decomposition of bilateral trade growth between Cambodia and its trading partners by region.** <u>Table 6</u> presents the decomposition of the Cambodian trade expansion with its top 30 trading partners across the six regions between 1993 and 2019. The analysis focuses on three key factors contributing to this growth: income, trade costs, and multilateral resistance. We organize the region in descending order based on its relative trade volumes with

Region	Due to GDP growth (%)	Due to the decline in trade costs (%)	Due to the decline in multilateral resistance (%)	Total (%)
Southeast Asia	+113.89	+27.30	-41.18	= 100
East Asia	+64.52	+82.73	-47.25	= 100
North America	+40.16	+43.48	+16.37	= 100
Europe	+64.22	+52.70	-16.92	= 100
South Asia	+92.63	+24.38	-17.01	= 100
Oceania	+62.15	+45.86	-8.01	= 100
Average	+59.65	+56.69	-16.34	= 100

Table 6. Decomposition of bilateral trade growth between Camboula and its trading partners by region between 1995 and 20	Table 6.	Decomposition	1 of bilateral trade	growth between	Cambodia a	nd its trading	partners by res	gion between	1993 and 20
--	----------	---------------	----------------------	----------------	------------	----------------	-----------------	--------------	-------------

Notes: The sum of the GDPs of each pair in 2019 is used as a weight to calculate the group's average in each region and the whole sample. The base-year data are computed as the three-year average from 1993 to 1995 to avoid annual volatility during Cambodia's economic reforms to a market-oriented economy in 1993 and to establish a balanced sample for all 30 trading partners—source: Authors' calculation using Eq (16).

https://doi.org/10.1371/journal.pone.0311754.t006

Trading Partner	Growth in trade (%)	Due to GDP growth (%)	Due to the decline in trade costs (%)	Due to the decline in multilateral resistance (%)	Total (%)
China	1,145.36	+73.87	+65.55	-39.42	= 100
United States	1,099.13	+39.95	+40.72	+19.33	= 100
Thailand	477.28	+101.23	+33.48	-34.71	= 100
Vietnam	708.06	+108.76	+44.30	-53.06	= 100
Japan	578.07	+39.22	+121.64	-60.86	= 100
Germany	736.74	+44.66	+48.95	+6.39	= 100
United Kingdom	794.18	+49.13	+119.1	-68.22	= 100
Canada	1030.6	+42.72	+76.84	-19.56	= 100
Republic of Korea	741.68	+63.73	+115.71	-79.44	= 100
Singapore	394.18	+139	+26.55	-65.54	= 100

Table 7. Decomposition of bilateral trade growth between Cambodia and its ten largest trading partners between 1993 and 2019.

Notes: The base-year data are computed as the three-year average from 1993 to 1995 to avoid annual volatility during Cambodia's economic reforms to a marketoriented economy in 1993 and to establish a balanced sample for all 30 trading partners—source: Authors' calculation using Eq (16).

https://doi.org/10.1371/journal.pone.0311754.t007

Cambodia as of 2019. The finding illustrates that the growth of the bilateral trade between Cambodia and its major trading partners has been attributed mainly to the rise in income and the decline in trade costs. The average contribution rate of the income growth to the bilateral trade growth is +59.65%, which is slightly higher than the contribution of the decline in trade costs (+56.69%). With an average contribution rate of -16.34%, the decrease in multilateral resistance impedes the growth of the Cambodian trade.

Because of the expanding overall scale of the GDP, there have been significant increases in the bilateral trade between Cambodia and the six regions: Southeast Asia, East Asia, South Asia, Oceania, Europe, and North America. Among these regions, the growth of the Cambodian trade with Southeast Asia and South Asia has been driven mainly by the rise in income, which accounts for over 90% of the total. The contribution from the GDP growth of the other three regions, including East Asia, Europe, and Oceania, is also substantial, exceeding 60%. Although the total income growth has a lower influence on trade growth in North America than in other regions, it still amounts to 40.16%.

These geographic differences reflect that the economic growth of Southeast Asian countries has been the most crucial factor in the expansion of trade with Cambodia, with individual countries' contributions ranging from 101.23% in Thailand to 139% in Singapore (see Table 7). Besides countries in Southeast Asia, the bilateral trade growth of Cambodia has also been attributed to the rise of the GDP of China (73.87% from Table 7) in East Asia (64.52%) and India in South Asia (92.63%). Southeast Asia's growing economies are primarily responsible for the increase in income that has fueled the expansion of their bilateral trade with Cambodia, followed by South Asia and East Asia. Since the GDP growth of developing countries in these regions outperforms most developed nations, there is great potential for expanding Cambodian trade in the future. This finding is consistent with previous studies showing the significance of economic growth in enhancing trade between countries [24, 62].

We can observe that the decline in trade costs has a more substantial influence on the expansion of Cambodia's trade with East Asia and North America than income growth. Among the six regions, the largest impact on promoting trade growth between Cambodia and East Asia trading partners comes from the decrease in trade costs, which accounts for 82.73%. The Cambodian trade volumes with Europe, North America, and Oceania have also increased by more than 40% due to the reduction in trade costs.

Overall, the decline in multilateral resistance has reduced the growth of the Cambodian trade. The average contribution rate of the decrease in multilateral resistance in the sample period is –16.34%. Compared to other regions, Cambodia's trade with Southeast Asia and East Asia is negatively impacted by the decrease in multilateral resistance, with the contribution rate reaching –40%. This finding reflects the ability of trading partners in Southeast Asia and East Asia to reduce trade costs with their trading partners. As a result, Cambodia may experience the consequences of the trade diversion, in which its trading partners choose to trade more with countries in Southeast Asia and East Asia rather than with Cambodia. For example, among the six regions, the decline in trade costs contributes the most to the bilateral trade growth between Cambodia and East Asia (+82.73%). However, these gains have been counterbalanced by the negative impact of a decline in multilateral resistance, bringing the overall contribution rate of a decrease in trade costs to +35.48% (+82.73%–47.25%). In contrast, the rise of North American multilateral resistance has promoted Cambodia's foreign trade.

**5.3.2 Decomposition of bilateral trade growth between Cambodia and its ten largest trading partners.** Table 7 decomposes the growth of the bilateral trade between Cambodia and its ten largest trading partners into three main factors: income, trade costs, and multilateral resistance. As of 2019, Cambodia's top ten trading partners, arranged in descending order, are China, the United States, Thailand, Vietnam, Japan, Germany, the United Kingdom, Canada, the Republic of Korea, and Singapore.

It is worth mentioning that the share of Cambodia's total trade volumes with each country exceeded two percent, and these ten countries accounted for almost 76% of Cambodia's total trade volumes with the world as of 2019, according to the IMF's DOTS database [72]. Over the period from 1993 to 2019, we can observe that the growth of the bilateral trade between Cambodia and China is the largest, expanding by 1,145.36%. Cambodia has also increased its bilateral trade with the United States and Canada by more than 1,000% among the top ten trading partners. It is important to note that China is Cambodia's largest trading partner and primary import market, while the United States is its major export market. As of 2019, China accounted for almost 24% of Cambodia's total trade volumes, followed by the United States at about 13%. The trade volumes of these two countries were approximately equivalent to the combined trade volumes of the eight remaining top 10 trading partners [72]. With its neighbors, Cambodia's trade has grown rapidly, with the top five largest economies in the world as of 2019 (the United States, China, Japan, Germany, and the United Kingdom), all of which have climbed by more than 500%.

It is evident that the primary factors driving Cambodia's bilateral trade growth are the rise in income and the decline in trade costs. From the perspective of the income contribution, the trade growth between Cambodia and Singapore is mainly due to income growth (139%). In comparison, the growth of trade between Cambodia and Japan because of income growth is just 39.22%. We can observe that the expansion of Cambodian bilateral trade with China, Singapore, Thailand, and Vietnam is primarily driven by the rise in income, which accounts for over 70% of the growth.

The contribution of the decline in trade costs is the highest between Cambodia and Japan, reaching 121.64%, followed by the United Kingdom (+119.1%) and the Republic of Korea (+-115.71%). The increase in Cambodia's trade with Canada, Germany, Japan, the Republic of Korea, the United Kingdom, and the United States is mainly explained by the decline in bilateral trade costs. Overall, the contribution rate of the decline in bilateral trade costs between Cambodia and other countries except for Japan, the United Kingdom, the Republic of Korea, Canada, and China is lower than 50%, ranging from +26.55% in Singapore to +48.95% in Germany.

Apart from the United States and Germany, Cambodia's bilateral trade with other major trading partners is diverted due to the decrease in multilateral resistance. The profound impact of multilateral resistance is seen in the Republic of Korea (-79.44%), followed by the United Kingdom (-68.22%), Singapore (-65.54%), Japan (-60.86%) and Vietnam (-53.06%). This finding suggests that there has been a substantial decrease in multilateral trade barriers between these countries and their trading partners, resulting in a significant trade diversion impact on Cambodia. For example, in the case of Vietnam, the decline in multilateral resistance led to a decrease in trade between Cambodia and Vietnam by -53.06%. Vietnam has made substantial progress in its trade openness with the world since its membership in the WTO in 2007. Vietnam's rapid elimination of trade barriers with its trading partners has also diverted trade away from Cambodia. Though the decline in the bilateral trade costs between Cambodia and Vietnam significantly boosts the bilateral trade between the two countries by +-44.30%, the adverse impact of multilateral resistance diminishes this positive driving force to – 8.76% (+44.30%-53.06%). Overall, the expansion of the bilateral trade between Cambodia and Vietnam has been driven mainly by income growth. Therefore, there will be the great potential for trade growth in the future as the two countries increase their living standards. In contrast, the impact of multilateral resistance turns positive for Germany and the United States, indicating that the increase in multilateral resistance will make Cambodia's trade attractive for these two countries.

# 6. Conclusions

# 6.1 Major findings

This study employs Novy's micro-founded method of trade costs to measure Cambodia's trade costs with its top 30 trading partners from 1993 to 2019. The study also decomposes the growth of the Cambodian trade during the same period into three driving forces: income, trade costs, and multilateral resistance. After the in-depth analyses, the following conclusions have been drawn:

Between 1993 and 2019, Cambodia's average trade costs decreased significantly by 35.43 percent, from 139.7 percent to 90.2 percent. There were fluctuations in its average trade costs until 2014, even though Cambodia joined the WTO in 2004. The remarkable 24.66 percent decline between 2014 and 2019 alone was primarily responsible for the overall decrease in the country's average trade costs between 1993 and 2019. Regarding the individual trading partners, Cambodia's bilateral trade costs with Hong Kong (China) and its neighboring countries (Thailand and Vietnam) have dropped to less than 50 percent. The bilateral trade costs of these three trading partners, along with Singapore, remained below 100 percent over the years from 1993 to 2019, which were consistently lower than those of other trading partners. In terms of the regions, Cambodia has enjoyed lower trade costs with its major trading partners in Southeast Asia and East Asia than with those in South Asia, Oceania, Europe, and North America.

As far as the magnitudes of the changes are concerned, all trading partners have reduced their trade costs, except for the Russian Federation. Thus, there is an opportunity for further optimization of trade relations between Cambodia and the Russian Federation if both countries can decrease their bilateral trade costs. Notably, China, Denmark, Hong Kong (China), Ireland, the Republic of Korea, Spain, Switzerland, and the United Kingdom experienced a remarkable decline in bilateral trade costs with Cambodia, exceeding 60 percent between 1993 and 2019. On the other hand, Finland, Germany, Malaysia, and the Netherlands all saw a decrease in trade costs of less than 20 percent. Among the top ten trading partners, Cambodia's bilateral trade costs with China, the United States, Thailand, Vietnam, Japan, the United

Kingdom, the Republic of Korea, and Singapore have declined to less than 100 percent. In contrast, Cambodia's bilateral trade costs with Germany and Canada have remained high, surpassing 100 percent. Furthermore, only Canada, Germany, and the United States had bilateral trade costs higher than the average among the top ten trading partners in 2019. Therefore, more attention is needed to enhance trade facilitation and improve the trade structure among these three trading partners.

In terms of trade costs among different groups, Cambodia's average trade costs with developing and emerging economies are lower than those with developed economies. Since 2014, Cambodia's average trade costs with developing and emerging economies have declined at double the rate of those with developed economies. Between 2014 and 2019, average trade costs between Cambodia and the trading partners along the Belt and Road Initiative (BRI) corridors decreased by 34.78 percent, twice as fast as observed with the non-BRI trading partners. These findings demonstrate that improvements in trade facilitation have led to a gradual decline in Cambodia's average trade costs, primarily attributed to the increasing investment in infrastructure projects in Cambodia and its major trading partners along the BRI corridors.

Regarding the decomposition analysis of trade growth, we find that the expansion of Cambodia's trade between 1993 and 2019 was driven by the rise in income (59.65 percent), the decline in trade costs (56.69 percent), and the decline in multilateral resistance (–16.34 percent). The expansion of trade opportunities among Cambodia's major trading partners and global markets led to a decline in their respective multilateral resistance, which diverted Cambodia's trade growth by more than 16 percent.

#### 6.2 Policy implications and recommendations

This study will contribute to a better understanding of Cambodia's development process and endeavors toward its trade integration with the world over the past two decades, starting from the initial stages of its economic reforms in the early 1990s until the recent period. Moreover, these findings have practical implications in providing new insights for formulating trade strategies to support Cambodia's ambition to transition from a least-developed country to an upper-middle-income country by 2030 and eventually progress towards a high-income country by 2050.

Our findings suggest that Cambodia has the potential to optimize its trade growth by focusing on its economic diplomacy strategies with the trading partners exhibiting high economic growth potential and those showing substantial reductions in trade costs. Over the past two decades, Cambodia has made considerable progress in its integration into the global economy through its membership in the ASEAN in 1999 and the WTO in 2004. As a result, the country's overall trade costs have been decreasing steadily. However, these costs have remained higher than those in major developed and newly industrialized countries. First, Cambodia should explore the possibility of reducing its bilateral trade costs among its major trading partners, especially those countries that have experienced high trade costs. For instance, major emerging economies, such as India, Indonesia, and the Russian Federation, demonstrate huge economic potential. However, their bilateral trade costs with Cambodia have remained high, thereby hindering the potential for trade expansion. Thus, it is essential to focus on ongoing efforts to mitigate trade frictions and improve trade facilitation among these countries.

Second, Cambodia should reduce its bilateral trade costs with major import trading partners that supply the necessary goods for the country's production, such as intermediate goods, raw materials, and capital goods. To achieve these goals, the government can implement a series of measures. (i) Cambodia should offer tariff exemptions on intermediate goods, raw materials, and capital goods, thus incentivizing efficient production processes. (ii) The government should continue to adopt the opening-up policies that foster international trade and economic cooperation. These strategies can be achieved by seeking the opportunity to sign more regional trade agreements (RTAs) or economic partnership agreements with its major trading partners. The aim is to promote mutually beneficial trade cooperation and enhance the country's integration into global value and supply chains. (iii) The government should prioritize the investment in infrastructure development. The effective investment strategy should involve establishing economic zones, free-trade zones, free-trade ports, industrial parks, and logistic hubs near borders and ports. Implementing this measure is significant in facilitating the development of industrial clusters that bring together diverse enterprises, thereby minimizing trade costs associated with shipping goods. (iv) The government should eliminate the different obstacles created by various institutions for transporting goods across borders, which are imperative for reducing trade costs.

In light of the current backlash against globalization, Cambodia should advocate for more openness for international trade and support regional and global efforts to promote trade facilitation between countries or regions.

# 6.3 Limitations of the study

Despite providing significant policy insights into the scales, evolutions, and impacts of Cambodian trade costs over the past two decades, this study has certain caveats. First, this paper only examines Cambodia's trade costs with its top 30 trading partners. Measuring Cambodia's average trade costs with all its trading partners is necessary to comprehensively understand the evolutions of the country's aggregate trade barriers over time. However, the limited data availability makes it impossible to determine Cambodia's aggregate trade costs with all its trading partners. Second, this research does not cover the most recent economic partnership agreements and regional trade agreements (RTAs), including the Cambodia-Hong Kong (China) Free Trade Agreement in February 2021 under the ASEAN-Hong Kong (China) Free Trade Agreement (AHKFTA), the Cambodia-China Free Trade Agreement (CCFTA) in January 2022, the Regional Comprehensive Economic Partnership (RCEP) in January 2022, the Cambodia-Republic of Korea Free Trade Agreement (CKFTA) in December 2022, and the Cambodia-United Arab Emirates Comprehensive Economic Partnership Agreement (CAM-UAE CEPA) in 2023, which are already in force. With the implementation of these agreements, Cambodia can anticipate a significant reduction in its average trade costs.

Furthermore, our analysis focuses solely on total trade costs in the goods sector and does not account for disaggregated trade costs within different industries. Finally, it is important to acknowledge the limitations inherent in the modeling approaches, which are based on several assumptions. As a result, the actual trade costs that Cambodia has experienced in practice may differ from the estimates generated by the models. Therefore, the findings should be interpreted with caution.

# 6.4 Future research directions

This work can serve as a benchmark for future studies, as it is the first research endeavor to measure Cambodian trade costs. Several areas require further investigation. (i) Expanding the sample size will provide a more thorough understanding of trade costs from a diverse range of countries (both developed and developing countries) in different regions. This will contribute to formulating trade policies that are well-suited to the specific circumstances of countries, considering the disparities in trade costs among countries and regions. (ii) In recent years, the Royal Government of Cambodia (RGC) has made efforts to sign, negotiate, and propose multiple RTAs and economic partnership agreements with different trading partners. In addition,

there has been a significant amount of investment in infrastructure, aiming to reduce trade costs. For example, Cambodia recently commenced the groundbreaking construction of the "Funan Techo Canal", spanning over 180 kilometers, on August 5, 2024. Given that this mega-project is imperative to Cambodia's waterway transport infrastructure, its completion within five years will significantly reduce the country's overall trade costs. Therefore, the determinants influencing trade costs, such as the current RTAs and infrastructure development, should also be explored. (iii) For a better understanding of Cambodia's participation in global value chains (GVCs), measuring trade costs using trade in value-added should be considered in future research. (iv) Investigating trade costs associated with various categories of goods or industries with high potential for growth, including agriculture, manufacturing, and services, requires attention in the future. This could yield findings that would be valuable for researchers, scholars, and policymakers.

# Supporting information

**S1 Appendix. Details of the data descriptions.** (DOCX)

# Acknowledgments

The authors would like to express their gratitude to the editors and the four anonymous reviewers for their critical and constructive comments as well as insightful suggestions, which significantly contributed to improving the quality of the earlier version of the manuscript.

# **Author Contributions**

**Conceptualization:** Borin Keo, Bin Li.

Data curation: Borin Keo.

Formal analysis: Borin Keo.

Investigation: Borin Keo.

Methodology: Borin Keo.

Project administration: Borin Keo.

Resources: Borin Keo.

Software: Borin Keo.

Supervision: Bin Li.

Validation: Borin Keo.

Visualization: Borin Keo.

Writing - original draft: Borin Keo.

Writing - review & editing: Borin Keo, Waqas Younis.

# References

- 1. Frankel J, Romer D. Trade and Growth: An Empirical Investigation. Cambridge, MA: National Bureau of Economic Research; 1996 Mar. Report No.: w5476. https://doi.org/10.3386/w5476
- Frankel JA, Romer D. Does Trade Cause Growth? American Economic Review. 1999; 89: 379–399. https://doi.org/10.1257/aer.89.3.379

- Jacks DS, Meissner CM, Novy D. Trade Costs, 1870–2000. American Economic Review. 2008; 98: 529–534. https://doi.org/10.1257/aer.98.2.529
- 4. Jacks DS, Meissner CM, Novy D. Trade booms, trade busts, and trade costs. Journal of International Economics. 2011; 83: 185–201. https://doi.org/10.1016/j.jinteco.2010.10.008
- Basile R, Commendatore P, De Benedictis L, Kubin I. The impact of trade costs on the European Regional Trade Network: An empirical and theoretical analysis. Review of International Economics. 2018; 26: 578–609. https://doi.org/10.1111/roie.12314
- Miroudot S, Ragoussis A. Vertical Trade, Trade Costs and FDI. Paris: OECD Publishing; 2009 Jul. Report No.: 89. https://doi.org/10.1787/222111384154
- Neary JP. Trade costs and foreign direct investment. International Review of Economics & Finance. 2009; 18: 207–218. https://doi.org/10.1016/j.iref.2008.06.004
- Venables AJ. Integration and the export behaviour of firms: Trade costs, trade volumes and welfare. Weltwirtschaftliches Archiv. 1994; 130: 118–132. https://doi.org/10.1007/BF02706012
- 9. Anderson JE, van Wincoop E. Borders, Trade, and Welfare. Engel C, Freund C, editors. Brookings Trade Forum. 2002; 207–243.
- Anderson JE, van Wincoop E. Trade Costs. Journal of Economic Literature. 2004; 42: 691–751. https:// doi.org/10.1257/0022051042177649
- 11. Hummels D, Ishii J, Yi K-M. The nature and growth of vertical specialization in world trade. Journal of International Economics. 2001; 54: 75–96. https://doi.org/10.1016/S0022-1996(00)00093-3
- 12. OECD/WTO. Aid for Trade at a Glance 2015: Reducing Trade Costs for Inclusive, Sustainable Growth. WTO, Geneva/OECD Publishing, Paris; 2015. https://doi.org/10.1787/aid\_glance-2015-en
- Diakantoni A, Escaith H, Roberts M, Verbeet T. Accumulating Trade Costs and Competitiveness in Global Value Chains. Geneva: World Trade Organization (WTO); 2017 Jan. Report No.: ERSD-2017-02.
- World Bank. Cambodia Product Imports 2021. In: World Integrated Trade Solution (WITS) database [Internet]. 2023 [cited 4 Aug 2023]. Available: https://wits.worldbank.org/CountryProfile/en/Country/ KHM/Year/2021/TradeFlow/Import/Partner/WLD/Product/all-groups,
- Ministry of Commerce (MoC). Cambodia Trade Integration Strategy 2019–2023. Ministry of Commerce (MoC); 2019.
- 16. Royal Government of Cambodia (RGC). Pentagonal Strategy-Phase I for Growth, Employment, Equity, Efficiency and Sustainability: Building the Foundation Towards Realizing the Cambodia Vision 2050 of the Royal Government of Cambodia of the Seventh Legislature of the National Assembly. Phnom Penh: Royal Government of Cambodia (RGC); 2023 Aug.
- Deardorff AV. Local comparative advantage: Trade costs and the pattern of trade. International Journal of Economic Theory. 2014; 10: 9–35. https://doi.org/10.1111/ijet.12025
- Dennis A, Shepherd B. Trade Costs, Barriers to Entry, and Export Diversification in Developing Countries. Washington, D.C: The World Bank; 2007. Report No.: WPS4368. <u>https://doi.org/10.1596/1813-9450-4368</u>
- Mora J, Olabisi M. Economic development and export diversification: The role of trade costs. International Economics. 2023; 173: 102–118. https://doi.org/10.1016/j.inteco.2022.11.002
- Yi K-M. Can Multistage Production Explain the Home Bias in Trade? American Economic Review. 2010; 100: 364–393. https://doi.org/10.1257/aer.100.1.364
- Jacks DS, Meissner CM, Novy D. Trade costs in the first wave of globalization. Explorations in Economic History. 2010; 47: 127–141. https://doi.org/10.1016/j.eeh.2009.07.001
- 22. Kim S. An Analysis of Cambodia's Trade Flows: A Gravity Model. MPRA Paper. University Library of Munich, Germany; 2006. Report No.: 21461. Available: https://ideas.repec.org/p/pra/mprapa/21461. html
- Huot N, Kakinaka M. Trade Structure and Trade Flows in Cambodia: A Gravity Model. ASEAN Economic Bulletin. 2007; 24: 305–319.
- Novy D. Gravity redux: measuring international trade costs with panel data. Economic Inquiry. 2013; 51: 101–121. https://doi.org/10.1111/j.1465-7295.2011.00439.x
- Duval Y, Utoktham C. Intraregional trade costs in Asia: a primer. Asia-Pacific Development Journal. 2012; 18: 1–23. https://doi.org/10.18356/e42f7c1f-en
- Duval Y, Saggu A, Utoktham C. REDUCING TRADE COSTS IN ASIA-PACIFIC DEVELOPING COUN-TRIES. Bangkok: The United Nations Economic and Social Commission for Asia and the Pacific (UN. ESCAP); 2015. Report No.: No. 84.
- Arvis J-F, Duval Y, Shepherd B, Utoktham C, Raj A. Trade Costs in the Developing World: 1996–2010. World Trade Review. 2016; 15: 451–474. https://doi.org/10.1017/S147474561500052X

- Novy D. Is the Iceberg Melting Less Quickly? International Trade Costs after World War II. University of Warwick; 2006. Report No.: No 764.
- Gaurav A, Mathur SK. Determinants of Trade Costs and Trade Growth Accounting between India and the European Union during 1995–2010. The World Economy. 2016; 39: 1399–1413. https://doi.org/10. 1111/twec.12317
- Xu T, Liang X. Measuring aggregate trade costs and its empirical effects on manufacturing export composition in China. China Finance and Economic Review. 2017; 5: 6. https://doi.org/10.1186/s40589-017-0049-z
- **31.** Milner C, Mcgowan D. Trade Costs and Trade Composition. Economic Inquiry. 2013; 51: 1886–1902. https://doi.org/10.1111/j.1465-7295.2012.00496.x
- Egger P, Larch M, Nigai S, Yotov Y. Trade Costs in the Global Economy: Measurement, Aggregation and Decomposition. Geneva: World Trade Organization (WTO); 2021. Report No.: ERSD-2021–2. https://doi.org/10.30875/e6c4c0b1-en
- Miroudot S, Sauvage J, Shepherd B. Measuring the cost of international trade in services. World Trade Review. 2013; 12: 719–735. https://doi.org/10.1017/S1474745613000049
- Wen S, Zheng J, Liu X. An analysis on China's agricultural bilateral trade costs? 1995-2007. China Agricultural Economic Review. 2013; 5: 360–372. <u>https://doi.org/10.1108/CAER-10-2011-0134</u>
- Turkson FE. How well does observable trade data measure trade friction costs? Evidence from member countries within the Economic Community of West African States (ECOWAS). Journal of African Trade. 2018; 5: 69–86. https://doi.org/10.1016/j.joat.2018.03.001
- de Soyres F, Mulabdic A, Murray S, Rocha N, Ruta M. How Much Will the Belt and Road Initiative Reduce Trade Costs? Washington, DC: World Bank; 2018 Oct. https://doi.org/10.1596/1813-9450-8614
- 37. Hang CN. Trade liberalisation: a Cambodian perspective. The Australian National University; 2003.
- WTO. WTO Stats. In: World Trade Organization (WTO) [Internet]. 2022 [cited 17 Sep 2022]. Available: https://stats.wto.org
- World Bank. Merchandise trade (% of GDP)—Cambodia. In: World Development Indicators Database [Internet]. 2022 [cited 4 Aug 2022]. Available: https://data.worldbank.org/indicator/TG.VAL.TOTL.GD. ZS?end=2019&locations=KH&start=1993
- WTO. World Trade Statistical Review 2017. Geneva: World Trade Organization; 2017. <u>https://doi.org/10.30875/dbeb8c4d-en</u>
- World Bank. Resilient Development—A Strategy to Diversify Cambodia's Growth Model: Cambodia Country Economic Memorandum. Phnom Penh: World Bank; 2021. Available: <u>https://openknowledge.worldbank.org/handle/10986/36908</u>
- 42. WTO and EIF. Trade impacts of LDC graduation: Cambodia. Geneva: World Trade Organization (WTO) & Enhanced Integrated Framework (EIF); 2022.
- WTO. World Trade Report 2014: Trade and Development: Recent Trends and the Role of the WTO. World Trade Organization (WTO); 2014. https://doi.org/10.30875/9f925144-en
- Bekkers E, Cariola G. The impact of LDC graduation on trade: A quantitative assessment. Geneva: World Trade Organization (WTO); 2022. Report No.: ERSD-2022-5. Available: <a href="https://www.wto-ilibrary.org/content/papers/10.30875/25189808-2022-5">https://www.wto-ilibrary.org/content/papers/10.30875/25189808-2022-5</a>
- 45. UNCTAD. A new centre of gravity: The regional comprehensive economic partnership and its trade effects. Geneva: UNCTAD; 2021 Dec. Report No.: UNCTAD/DITC/INF/2021/5. Available: https://unctad.org/system/files/official-document/ditcinf2021d5\_en\_0.pdf
- Samuelson PA. The Transfer Problem and Transport Costs, II: Analysis of Effects of Trade Impediments. The Economic Journal. 1954; 64: 264–289. https://doi.org/10.2307/2226834
- Deardorff AV. Testing trade theories and predicting trade flows. In: Jones RW, Kenen PB, editors. Handbook of International Economics. Elsevier; 1984. pp. 467–517. <u>https://doi.org/10.1016/S1573-4404(84)01013-3</u>
- Krugman P. Increasing returns, monopolistic competition, and international trade. Journal of International Economics. 1979; 9: 469–479. https://doi.org/10.1016/0022-1996(79)90017-5
- Krugman P. Scale Economies, Product Differentiation, and the Pattern of Trade. The American Economic Review. 1980; 70: 950–959.
- Fujita M, Krugman P, Venables AJ. The Spatial Economy: Cities, Regions, and International Trade. The MIT Press; 1999.
- Krugman P. Increasing Returns and Economic Geography. Journal of Political Economy. 1991; 99: 483–499. https://doi.org/10.1086/261763

- 52. Bernard AB, Eaton J, Jensen JB, Kortum S. Plants and Productivity in International Trade. American Economic Review. 2003; 93: 1268–1290. https://doi.org/10.1257/000282803769206296
- Melitz MJ. The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity. Econometrica. 2003; 71: 1695–1725.
- Obstfeld M, Rogoff K. The Six Major Puzzles in International Macroeconomics: Is There a Common Cause? NBER Macroeconomics Annual. 2000; 15: 339–390. https://doi.org/10.2307/3585403
- Limão N, Venables AJ. Infrastructure, Geographical Disadvantage, Transport Costs, and Trade. The World Bank Economic Review. 2001; 15: 451–479. https://doi.org/10.1093/wber/15.3.451
- Hummels D. Transportation Costs and International Trade in the Second Era of Globalization. Journal of Economic Perspectives. 2007; 21: 131–154. https://doi.org/10.1257/jep.21.3.131
- McCallum J. National Borders Matter: Canada-U.S. Regional Trade Patterns. The American Economic Review. 1995; 85: 615–623.
- Anderson JE, van Wincoop E. Gravity with Gravitas: A Solution to the Border Puzzle. The American Economic Review. 2003; 93: 170–192. https://doi.org/10.1257/000282803321455214
- Chen N, Novy D. On the measurement of trade costs: direct vs. indirect approaches to quantifying standards and technical regulations. World Trade Review. 2012; 11: 401–414. https://doi.org/10.1017/ S1474745612000183
- **60.** Tinbergen J. Shaping the world economy: suggestions for an international economic policy. New York: Twentieth Century Fund; 1962.
- Rose AK. One money, one market: the effect of common currencies on trade. Economic Policy. 2000; 15: 8–0. https://doi.org/10.1111/1468-0327.00056
- Baier SL, Bergstrand JH. The growth of world trade: tariffs, transport costs, and income similarity. Journal of International Economics. 2001; 53: 1–27. https://doi.org/10.1016/S0022-1996(00)00060-X
- 63. Hummels D. Toward a Geography of Trade Costs. Purdue University; 1999 Jan.
- 64. Eaton J, Kortum S. Technology, Geography, and Trade. Econometrica. 2002; 70: 1741–1779.
- Wei S-J. Intra-National versus International Trade: How Stubborn are Nations in Global Integration? National Bureau of Economic Research; 1996. Report No.: 5531. https://doi.org/10.3386/w5531
- Head K, Ries J. Increasing Returns versus National Product Differentiation as an Explanation for the Pattern of U.S.-Canada Trade. American Economic Review. 2001; 91: 858–876. <u>https://doi.org/10.1257/aer.91.4.858</u>
- 67. Chaney T. Distorted Gravity: The Intensive and Extensive Margins of International Trade. American Economic Review. 2008; 98: 1707–1721. https://doi.org/10.1257/aer.98.4.1707
- Melitz MJ, Ottaviano GIP. Market Size, Trade, and Productivity. The Review of Economic Studies. 2008; 75: 295–316.
- Head K, Mayer T. Chapter 59—The Empirics of Agglomeration and Trade. In: Henderson JV, Thisse J-F, editors. Handbook of Regional and Urban Economics. Elsevier; 2004. pp. 2609–2669. https://doi. org/10.1016/S1574-0080(04)80016-6
- Krugman P. Growing World Trade: Causes and Consequences. Brookings Papers on Economic Activity. 1995; 1995: 327. https://doi.org/10.2307/2534577
- Feenstra RC. Integration of Trade and Disintegration of Production in the Global Economy. Journal of Economic Perspectives. 1998; 12: 31–50. https://doi.org/10.1257/jep.12.4.31
- 72. IMF's Direction of Trade Statistics (DOTS) database. In: International Monetary Fund (IMF) [Internet]. 25 Jan 2023. Available: https://data.imf.org/?sk=9D6028D4-F14A-464C-A2F2-59B2CD424B85
- 73. World Economic Outlook Database, October 2022. In: International Monetary Fund (IMF) [Internet]. 15 Oct 2022. Available: <u>https://www.imf.org/en/Publications/SPROLLs/world-economic-outlook-databases</u>
- Borchert I, Larch M, Shikher S, Yotov YV. The International Trade and Production Database for Estimation (ITPD-E). International Economics. 2021; 166: 140–166. https://doi.org/10.1016/j.inteco.2020.08. 001
- Borchert I, Larch M, Shikher S, Yotov YV. The International Trade and Production Database for Estimation—Release 2 (ITPD-E-R02). Washington, DC: U.S. INTERNATIONAL TRADE COMMISSION; 2022 Jul. Report No.: 2022–07–A. Available: https://www.usitc.gov/publications/332/working\_papers/ itpd\_e\_r02\_usitc\_wp.pdf
- IMF. Fiscal Monitor: Helping People Bounce Back. Washington, DC: International Monetary Fund (IMF); 2022. https://doi.org/10.5089/9798400212741.089
- Matzner A, Meyer B, Oberhofer H. Trade in times of uncertainty. The World Economy. 2023; 46: 2564– 2597. https://doi.org/10.1111/twec.13463

- **78.** Hamadeh N, Rompaey CV, Metreau E. World Bank Group country classifications by income level for FY24 (July 1, 2023- June 30, 2024). In: World Bank [Internet]. 30 Jun 2023. Available: https://blogs. worldbank.org/opendata/new-world-bank-group-country-classifications-income-level-fy24
- 79. Kotoski K, Chandara S. Cambodia asks EU to give its bicycle makers a break. In: Phnom Penh Post [Internet]. 7 Mar 2016 [cited 20 Sep 2022]. Available: https://www.phnompenhpost.com/business/ cambodia-asks-eu-give-its-bicycle-makers-break
- China to host 3rd Belt and Road Forum for Int'l Cooperation in October. In: Xinhua News Agency [Internet]. 31 Aug 2023 [cited 1 Sep 2023]. Available: https://english.news.cn/20230831/ 89fde063bed246cba6e115e272cfb36e/c.html
- 81. Chheang V. Cambodia Embraces China's Belt and Road Initiative. ISEAS-Yusof Ishak Institute; 2017. Report No.: 48. Available: https://think-asia.org/handle/11540/7213
- Chin H, He W. The Belt and Road Initiative: 65 Countries and Beyond. Hong Kong, China: Fung Business Intelligence Centre; 2016 May. Available: https://www.fbicgroup.com/internal-site/wp-content/ uploads/documents/BR\_Initiative\_65\_Countries\_and\_Beyond.pdf
- Chen Z, Li X. Economic impact of transportation infrastructure investment under the Belt and Road Initiative. Asia Europe Journal. 2021; 19: 131–159. <u>https://doi.org/10.1007/s10308-021-00617-3</u> PMID: 34305499
- Dumor K, Yao L, Ainam J-P, Amouzou EK, Ayivi W. Quantitative Dynamics Effects of Belt and Road Economies Trade Using Structural Gravity and Neural Networks. SAGE Open. 2021; 11: 21582440211032662. https://doi.org/10.1177/21582440211032662
- **85.** OECD. The Belt and Road Initiative in the global trade, investment and finance landscape. Paris: OECD; 2018 Sep pp. 61–101. https://doi.org/10.1787/bus\_fin\_out-2018-6-en